The Vault

https://prism.ucalgary.ca

Open Theses and Dissertations

2023-01-27

# Phoretic Phenomena of Colloidal Particles in a Dead-end Pore Microfluidic Channel

## Nooryani, Matina

Nooryani, M. (2023). Phoretic phenomena of colloidal particles in a dead-end pore microfluidic channel (Master's thesis, University of Calgary, Calgary, Canada). Retrieved from https://prism.ucalgary.ca. http://hdl.handle.net/1880/115802 Downloaded from PRISM Repository, University of Calgary

## UNIVERSITY OF CALGARY

Phoretic phenomena of colloidal particles in a dead-end pore microfluidic channel

by

Matina Nooryani

## A THESIS

## SUBMITTED TO THE FACULTY OF GRADUATE STUDIES

## IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE

## DEGREE OF MASTER OF SCIENCE

## GRADUATE PROGRAM IN CHEMICAL ENGINEERING

CALGARY, ALBERTA

JANUARY, 2023

© Matina Nooryani 2023

## Abstract

Colloidal particles can be manipulated under the gradient of an external field to transport them in the system toward desired locations. The motion of colloidal particles resulting from an imposed or self-generated concentration gradient is called diffusiophoresis which has been used in a wide range of applications such as drug delivery, wastewater treatment, and enhanced oil recovery. This phenomenon provides a low-cost and easy portable method to manipulate particles. Previous works investigated the diffusiophoretic motion of non-porous particles in microfluidic channels both experimentally and theoretically. However, the migration of porous particles under the effect of concentration gradient has not been studied experimentally. In this work, we have used a deadend pore microfluidic channel to explore the diffusiophoretic motion of porous particles. For this purpose, we first study the migration of non-porous particles in the dead-end pore channel with and without the presence of background concentration gradients. Our results indicate a temporary self-generated concentration gradient is generated in the channel resulting from the interfacial interaction of PDMS walls with the ions present in DI water without introducing any background concentration gradient. In the presence of an electrolyte concentration gradient, the relative difference in the ions' diffusivity ( $\beta$ ) and particles' zeta potential are two main effective factors on the diffusiophoretic motion of non-porous particles. In the next section, we study the diffusiophoretic motion of porous particles, and differences in the case of non-porous particles are investigated. In contrast to non-porous particles, porous particles' migration under the influence of concentration gradient is a function of pore size, surface area, and surface charge density. The surface charge density of porous particles can change significantly because of counterions diffusion into the porous particles which can result in inversion of motion direction. Therefore, the effect of the interior microstructure of colloidal particles is crucial in controlling their diffusiophoretic motion.

## Acknowledgments

I would like to express my appreciation to the following people who helped and pushed me during my masters degree at the University of Calgary. Starting with Dr. Giovanniantonio Natale, my supervisor, who kept encouraging me to think deeply about the physics behind the research as a first step, and who kept pushing me to explore and follow new ideas. Also, I would like to thank my co-supervisor, Dr. Anne Benneker, who kept supporting and guiding me when I was stressed out and facing a problem in my research, and who was always there to remind me to be passionate and optimistic. I would like to thank them both for their insightful comments, encouragement, and supports me which turn me into the researcher I am today.

I thank my friends and colleagues from the Complex Fluid Lab and MicroTransport Lab for the help and support they have provided during my research. Thank you Mansoureh, Razieh, Narges, Amirreza, and others who helped me out whenever I needed anything. I would also like to thank CFREF for financial support during my masters degree.

I would like to thank my family, especially my husband, who continuously encouraged me and reminded me of my abilities. Also, I would like to thank my parents for all of their support and for believing in me.

# **Table Of Contents**

Abstract	ii	
Acknowledg	mentsiii	
List Of Figu	resvi	
List of Tabl	esix	
Nomenclatu	re x	
Chapter 1		
Introduction1		
1.1 Mo	tivation2	
1.2 Ob	jectives	
1.3 Tas	ks	
1.4 Org	ganization4	
Chapter 2		
2 Backgro	ound and Literature Review5	
2.1 Pho	pretic motion of colloidal particles	
2.2 Mie	crofluidics	
2.3 Dif	fusiophoresis of colloidal particles in electrolyte solutions	
2.3.1	Diffusiophoresis in a microfluidic device 11	
2.3.2	Diffusiophoresis in a dead-end pore channel12	
<b>2.3.3</b> PDMS interaction with the dissolved CO <sub>2</sub> in a solution in the microfluidic channel 16		
2.4 Eff	ect of the interior structure of colloids on their diffusiophoretic motion 17	
Chapter 3		
3 A self-g	enerated exclusion zone in a dead-end pore microfluidic channel	
3.1 Int	roduction	
3.2 Exp	perimental setup	
3.2.1	PDMS Microfluidic Channel	
3.2.2	Experimental Protocol	
3.3 Res	ults and discussion	
3.3.1	Exclusion Zone in the dead-end pore geometry	
3.3.2	Ion's boundary layer analysis	
3.3.3	Particles' trajectory analysis	

3.3.4	Evolution of the Exclusion zone		
3.3.5	Balance between diffusiophoretic velocity and convective velocity		
3.4 Con	nclusion		
Chapter 4			
4 The diff	usiophoretic motion of porous and non-porous silica particles in a dead-end pore		
microfluidio	channel		
4.1 Int	roduction		
4.2 Exp	perimental setup		
4.2.1	PDMS microfluidic channel		
4.2.2	Salt solutions and particle suspension		
4.2.3	Experimental protocol for microfluidic channel 40		
4.2.4	Experimental protocol for drag force measurement 40		
4.3 Res	sults and discussion		
4.3.1	Drag force measurement using optical tweezers		
4.3.2	Diffusiophoretic motion of non-porous particles		
4.3.3	Diffusiophoretic motion of porous particles		
4.4 Co	nclusions		
Chapter 5			
5 Conclus	sions and Recommendations55		
5.1 Cor	nclusions		
5.2 Rec	commendations and Future Directions58		
Supplement	ary information for chapter 3		
A 1. Exclu	ision zone establishment over time60		
A 2. Obta	ining ions' exclusion zone thickness using boundary layer analysis		
A 3. Obta	ining particles' exclusion zone thickness using particles' trajectory analysis 64		
A 4. Exclusion zone development in the main channel for different flow rates			
Reference			

## **List Of Figures**

Fig 2.1 Schematic representation of a charged particle under a concentration gradient in an electrolyte solution (NaCl). An electric field (E) is generated by the concentration gradient of NaCl, and a tangential pressure gradient in the electric double layer of the particle is produced due to the gradient of NaCl concentration. For the case of negative  $\beta$  like NaCl, both chemiphoresis and electrophoresis contribution drives the negatively charged particle towards higher ionic concentration. Reproduced from ref. 58 with permission from The Royal Society of Chemistry. 6 Fig 2.2 Schematic illustration of fabricating microfluidic channel using soft lithography technique. First, the degassed PDMS is poured on the silicon wafer. Second, after degassing the poured PDMS on the silicon wafer, the PDMS will be cured in the oven at 90 °C for 2 hours. Then, the cured PDMS can be peeled off and sealed to a glass slide. Reproduced from ref. 75 with permission Fig 2.3 A schematic illustration of the experimental setup to visualize the phoretic motion of colloidal particles under a microscope. This figure includes two syringe pumps to inject the solution into a microchip, connecting tubes, a microchip, and a microscope to visualize the particles' motion. Adapted with permission from ref. 77. Copyright 2006 American Chemical Fig 2.4 Sketch of a set of dead-end pores in the microfluidic channel. The upper section is the main channel and the small branches are called dead-end pores. Adapted with permission from ref. 29. Fig 2.5 Generating concentration gradient through the compaction experiment in a dead-pore channel. Initially, the pores are filled with the colloidal suspension (solution 1). Then an air plug following with the second solution is injected into the channel. The starting time of the experiment is after the two solutions come into contact at the entrance of the first pore. Adapted with Fig 2.6 Creating concentration gradient using entrainment approach in a dead-end pore microfluidic channel. First, the pores are filled with the electrolyte solution (solution 1), then, the colloidal suspension (solution 2) is flown into the channel after evacuating the main channel. Adapted with permission from ref. 29. Copyright 2020 American Chemical Society......14 Fig 2.7 The trajectories of 1µm particles in a series of dead-end pores using the entrainment experiment approach with and without the presence of a concentration gradient. (a) In the absence of a concentration gradient in the pores, colloids will not be captured inside the pores and they just enter the entrance of the pores due to the hydrodynamic forces and flow geometry effect. (b) In the presence of an electrolyte concentration gradient (10mM of NaCl inside the pore and 0.1mM outside the pore), the particles migrate into the pores as a result of the diffusiophoretic velocity. The colloid tracks in the main channel are fainter than (a) due to the difference in the contrast settings during the image processing. Reproduced from ref. 67 with permission from The Royal Fig 2.8 Schematic illustration of the imposed velocities on the colloidal particle in the presence of a concentration gradient at the entrance of dead-end pores. The convective velocity originated from the ambient flow of the particle, the diffusiophoretic velocity is created due to the imposed Fig 3.1 Schematic configuration of the dead-end pore geometry and dimensions. After filling pores with DI water, a suspension of silica particles flows through the evacuated main channel and Fig 3.2 A schematic representation (a) and experimental observation (b) of the exclusion zone formation in the main channel caused by a self-generated concentration gradient from the deadend pores. (a) The blue dashed line represents the ionic exclusion zone, which is larger than the particle exclusion zone represented by the red line. A particle in the main channel experiences forces in the tangential direction as a result of the ion concentration gradient.Experimental observation of the exclusion layer formation in the dead-end channel for pore width =  $100\mu m$ . 25 Fig 3.3 The experimental data of normalized particle exclusion thickness multiple by the square root of velocity for 2µm silica particles. The orange square shows experimental data in Q=80µL/hr, Fig 3.4 Comparing the exclusion zone thickness as a function of position over the first pore calculated in theory with experimental results for 2µm silica particles. (a) For pore width=50µm. (b) For pore width=100µm. (c) For pore width=150µm. The red curve represents the ionic Fig 3.5 Normalized exclusion zone thickness over time for 2µm silica particles. (a) For pore Fig 3.6 Normalized exclusion zone thickness over time for 1µm silica particles. (a) For pore Fig 3.7 Effect of the particle size on the sustainability of the exclusion zone over time. This figure is plotted for the end-point ( $y = 150\mu m$ ) of the first pore. The red curve and black curve represent Fig 3.8 Effect of increasing flow rate on the maximum exclusion thickness based on the theoretical calculations and experimental observations for three different flow rates. The pore width in this diagram is 100µm. Dashed lines show the ionic exclusion thickness and solid lines represent 2 µm silica particles exclusion thickness. (a) maximum exclusion thickness for each flow rate. (b) Scaling the dimensionless exclusion thickness by the inverse of the square root of the Sh number. 

Fig 4.1 Schematic illustration of the dead-end pore geometry and dimensions. The pores are filled with solution 1 (electrolyte solution), then, solution 2 which is particle suspension in DI water is Fig 4.2 Drag force of 2 µm non-porous silica particles and porous silica particles as a function of velocity using the optical tweezers. Linear fits to represent Stokes' law are included (dashed lines). Also, the effect of the ratio of pore size to water molecule size is presented. The green line representes the non-porous drag force where the ratio of pore size to water molecule size is zero. The blue line represents 2 µm porous particles with 2 nm pores where the ratio of pore size to water molecule size is 7.4. The yellow line shows 2 µm porous particles with 100 nm pores where Fig 4.3 Schematic illustration of negatively charged particles under the electrolyte concentration gradient. In this figure, the pore is filled with the electrolyte which the diffusivity of negative ions is higher than positive ions, and the particle suspension without any electrolyte is passing over the entrance of the pore. At the starting point (t = 0), the particles just reach the entrance of the pores. After reaching the entrance of the pore, they will migrate toward the inside of the pore under the influence of electrolyte concentration gradient. The mechanism happening at the entrance of the pore around the negatively charged particle is shown in the right hand side for the NaCl electrolyte as an example, where the slower effective motion of Na<sup>+</sup> results in an electric field that drives the Fig 4.4 Entrainment of 1 µm non-porous silica particles into the pores in the presence of the electrolyte concentration gradient as a function of time. (a) Migrating of 1 µm non-porous silica particles under the NaCl concentration gradient. (b) The motion of 1 µm non-porous silica particles Fig 4.5 Dimensionless maximum position of 1 µm non-porous silica particles under the imposed electrolyte concentration gradient over time. The red squares and blue circles represent the Fig 4.6 Diffusiophoretic motion of porous silica particles under the concentration gradient of 10 mM NaCl. (a)Time sequence images for 1 µm porous silica particles with 2nm pores (b) Time Fig 4.7 Scaled maximum position of 1 µm non-porous and porous silica particles under the 10 mM NaCl concentration gradient over time. The orange circles, green triangles, and red squares Fig 4.8 Diffusiophoretic motion of porous silica particles under the concentration gradient of 10mM NaOH. (a) Time sequence images for 1 µm porous silica particles with 2 nm pore size. (b) 

# List of Tables

Table 4.1 Zeta potential measurements	of non-porous and porous	particles with DLS instrument
in DI water and 10 mM NaOH solution.		

# Nomenclature

$\boldsymbol{U}_{DP}$	diffusiophoretic velocity
$C(\mathbf{x},t)$	solute concentration
$D_{DP}$	diffusiophoretic mobility of particles
$\kappa^{-1}$	Debye length
a	radius of the particle
β	relative diffusivity difference of anions and cations
Е	electrical permittivity of the solution
$k_B$	Boltzmann constant
Т	absolute temperature
μ	viscosity of the solution,
Ζ	valence of the ions
е	elementary charge
$\zeta_p$	zeta potential of particles
E	electric field
Wcs	width of the cross-section
Re	Reynolds number
ρ	space charge density of mobile ions in the system
γ	friction coefficient representing the reduced drag force in a porous material,
р	fluid pressure
v	fluid flow
φ	electric potential
$\lambda^{-1}$	Brinkman screening length
<i>C</i> <sub>0</sub>	initial electrolyte concentration
$\bar{Q}$	dimensionless fixed-charge density,
$A_1$	mathematical function of permeability for the electrophoretic contribution

- $A_2$  mathematical function of permeability for the chemiphoretic contribution
- **n** outward normal vector
- $L_{ch}$  channel length
- Sh Sherwood number

## **Chapter 1**

## Introduction

The phoretic motion of colloidal particles which is the transport of colloidal particles under the influence of external fields has been investigated for over fifty years [1]-[3] due to its wide range of applications in separation[4]–[8], wastewater treatment[9]–[11], drug delivery[12]–[20], enhanced oil recovery (EOR)[21]–[23], and many others. The typical external fields are a gradient in electrostatic potential, solute concentration, or temperature. The motion of colloidal particles under the effect of a solute concentration gradient is called diffusiophoresis[24]-[26]. The mechanism of this phenomenon in the presence of electrolytes or non-electrolytes is different. In most industrial applications, the presence of ions is common. The diffusiophoretic motion of colloidal particles in electrolytes has been widely studied both theoretically[27], [28] and experimentally[29]–[31]. Experimental studies have utilized microfluidic devices to analyze the diffusiophoretic motion of colloidal particles because microfluidics provides practical advantages such as a cost-effective process, carrying out experiments in a short time, high resolution, etc[32]. Among different materials for fabricating microfluidic channels, polydimethylsiloxane (PDMS) is the most popular[29], [30], [33]-[36]. PDMS has high gas permeability and negative surface charge which both can affect the motion of charged colloidal particles by introducing a temporary concentration gradient. However, the influence of these effects on the diffusiophoretic motion of colloidal particles has been relatively unexplored in literature.

Furthermore, the diffusiophoretic transport of non-porous particles has been explored in various studies both theoretically and experimentally; however, in reality, many colloidal particles are not non-porous but of porous nature meaning that they are permeable to the liquid medium and ions. Porous particles such as DNA, proteins, porous silica, etc are broadly used in various fields such as drug delivery and water treatment[37]. Therefore, it is important to understand their diffusiophoretic motion as diffusiophoresis is a potential technique to use in drug delivery and water treatment applications. The diffusiophoretic motion of charged porous particles has been investigated theoretically in limited studies in the last decade[38]–[41]; whereas, there is no experimental study on the migration of porous particles under the effect of a concentration gradient. Since various parameters such as pore size, porosity (defined as the volume of the pores

to the total volume), how the pores are connected, Debye length, and others can affect the migration of porous particles, further experimental and theoretical investigations are required to be able to predict their motion under the influence of concentration gradient.

## 1.1 Motivation

The work explained in this thesis is arising from a gap in the literature on two experimental aspects of diffusiophoresis. The first one is the motion of charged colloidal particles in a microfluidic channel under the influence of the surface chemistry of PDMS and the interaction of the PDMS wall with the fluid inside the channel. Second, the motion of porous particles under the effect of a concentration gradient in a microfluidic channel which has not been investigated experimentally. Therefore, we aim to study the effects of PDMS surface chemistry and the influence of its interaction with an aqueous fluid on the particles' phoretic motion. A better understanding of the effect of PDMS surface chemistry and the effects of this chemistry on particle motion will yield insights and understanding in many experimental microfluidic devices where charged colloidal particles suspended in a solution are moving through a microfluidic channel.

Additionally, we experimentally investigate the diffusiophoretic motion of porous particles in a microfluidic channel, which has been discussed in detail for non-porous particles in numerical and experimental studies. Theoretically, the migration of charged porous particles under an imposed concentration gradient depends on the surface charge, porosity, Debye length, volume fraction of particles in the solution, and difference in the ions' diffusion coefficient[38]. In theoretical studies, it has been proved as permeability increases, the diffusiophoretic velocity increases as well[41]. The permeability of porous particles which means passing fluid through the inside of particles can change by the way that pores are connected, pore size, and porosity of the interior of porous particles. Increasing the permeability reduces the imposed drag force on porous particles resulting in a higher velocity[42]. Therefore, various factors affect the permeability of porous particles which is not considered in the previous studies[38]–[41], [43]. Also, the surface charge of porous particles as another effective parameter on the diffusiophoretic velocity[39] is very important because it changes by entering counterions into the interior region of porous particles. Understanding the phoretic motion of porous particles under a concentration gradient can be applied in particle manipulating applications. This motivated us to explore the effect of porosity

as a function of pore size, and the effective surface charge of porous particles on their diffusiophoretic behavior under an electrolyte concentration gradient experimentally, which has relatively ample theoretical evidence.

## **1.2 Objectives**

Based on the motivation for studying the motion of non-porous and porous colloidal particles in a microfluidic channel with and without the imposed concentration gradient, the following objectives are defined:

- 1. Investigate the influence of PDMS walls' interaction and surface chemistry with the aqueous fluid on the motion of colloidal particles in a microfluidic channel.
- Measure the migration of charged porous particles with different porosity in the presence of an electrolyte concentration gradient using a microfluidic chip and compare their diffusiophoretic motion with that of non-porous particles.

## 1.3 Tasks

To answer the above-mentioned objectives the following steps are taken into account through this research:

- The motion of non-porous particles will be visualized in a microfluidic channel to analyze the influence of PDMS interaction on the particles' migration in the channel along with considering the effect of flow velocity. Also, the effect of particle size on their motion will be observed and analyzed.
- The difference in hydrodynamic forces between porous and non-porous particles will be measured using an optical tweezers instrument.
- The diffusiophoretic motion of porous particles with different porosity in a microfluidic channel will be investigated and their motion under the concentration gradient of two different electrolytes with different ions diffusivity will be compared with non-porous particles.

## 1.4 Organization

This thesis is organized in the following manner: Chapter 2 provides a background and literature review on the diffusiophoretic motion of colloidal particles in microfluidic channels. Chapter 3 covers the details of the motion of colloidal particles in a microfluidic channel and the effect of PDMS wall interaction on particles' migration. Chapter 4 discusses the motion of porous and non-porous particles under the electrolyte concentration gradient in a microfluidic channel. Also, the drag force comparison of porous particles with non-porous particles measured by an optical tweezer is covered in this chapter. Chapter 5 provides a summary of the results and conclusion of the work done as well as recommendations for future studies on our topic.

## **Chapter 2**

## 2 Background and Literature Review

This chapter focuses on providing a brief literature review of investigations that have been done on the phoretic motion of colloidal particles under a concentration gradient, and the experimental studies of this phenomenon in a microfluidic channel. This chapter starts with an introduction to phoretic motion in general. Then, we provide an overview of microfluidics as the experimental work of this thesis has been done in a microfluidic channel. Afterwards, the phoretic motion of colloidal particles under a concentration gradient (diffusiophoresis) is explained in more detail in a solution, and in a specific microfluidic geometry that is relevant to our work. In the final section, the influence of the colloids' interior structure on their diffusiophoretic motion is elaborated. Also, the gaps related to this topic will be identified.

#### **2.1** Phoretic motion of colloidal particles

The migration of colloidal particles under imposed external fields is termed phoretic motion. The external field that leads to the transport of particles can originate from a gradient in solute concentration (diffusiophoresis), electrostatic potential (electrophoresis), or temperature (thermophoresis). The motion of particles under these gradients is produced by the interaction of the ambient fluid with the interfacial structure near the particles' surfaces[44]–[46]. Phoretic motion has two important characteristics: first, the origin of this motion is within the interfacial diffuse layer near the particle's surface which is the electric double layer for charged particles, and other surface interactions characterized by a diffuse layer due to the osmotic pressure[47], [48]. Second, the motion of the particle is force-free and the particle moves at a steady velocity[49]. In this thesis, we focused on the migration of colloidal particles under a concentration gradient of solute molecules.

The motion of colloidal particles resulting from an imposed or self-generated concentration gradient is called diffusiophoresis and has been widely used in various fields such as drug delivery,

wastewater treatment, separation, laundry detergency, and enhanced oil recovery [50]–[56]. In general, there are two types of diffusiophoresis in the presence of or without electrolytes. In electrolytes, the concentration gradient of solute molecules causes particle transport by the generation of a local electric field (electrophoresis) and pressure field in the Debye layer (chemiphoresis). Therefore, a combined influence of electrophoresis and chemiphoresis drives particles under gradients of electrolyte solution. The electrophoresis contribution leads to the particle motion due to the induced electric field as a result of a difference in the cations' and anions' diffusivities. Simultaneously, the chemiphoresis contribution produces the osmotic pressure gradient along a particle surface in the presence of local solute gradients which drives particles in the solution as shown in Fig 2.1[57], [58].



Fig 2.1 Schematic representation of a charged particle under a concentration gradient in an electrolyte solution (NaCl). An electric field (E) is generated by the concentration gradient of NaCl, and a tangential pressure gradient in the electric double layer of the particle is produced due to the gradient of NaCl concentration. For the case of negative  $\beta$  like NaCl, both chemiphoresis and electrophoresis contribution drives the negatively charged particle towards higher ionic concentration. Reproduced from ref. 58 with permission from The Royal Society of Chemistry.

The osmotic pressure gradient along the particle surface generates a fluid flow along the particle surface which is called diffusioosmosis. In non-electrolytes, the transport of particles is solely

through the contribution of chemiphoresis due to particle-solute interactions such as van der Waals interactions or steric exclusion[44], [57], [58].

### 2.2 Microfluidics

Microfluidics is a field of study of fabricating, designing, and utilizing of systems that deal with the transport and manipulation of small amounts of fluids (10<sup>-9</sup> to 10<sup>-18</sup> liters) in small channels with characteristic dimensions of tens to hundreds of micrometers. This technology offers new useful capabilities compared to large-scale protocols such as less sample or reagent consumption, detecting and controlling the experiment with high resolution and sensitivity, low-cost process, running experiments in a short time, reduced risk of contamination, laminar flow, and ability to carry out multiple processes in parallel[32], [59], [60]. In recent years, microfluidics has been widely used to manipulate the phoretic motion of particles under a gradient of electric potential[61]–[63], temperature[64]–[66], and concentration[29], [30], [52], [57], [67]–[69] as the particles' motion can be visualized with a high resolution.

The current common prototyping of microchannels involves fabricating patterns of the microfluidic channel on a substrate. A popular prototyping process is utilizing soft lithography procedures[70]. For using this method, a design of the channel is created in software such as CAD or KLayout which is printed on a photo mask. Then, the pattern will be printed on a master mold which is the silicon wafer. Afterwards, the master mold (silicon wafer) is silanized using trichlorosilane (HSiCl<sub>3</sub>) to create a hydrophobic surface on the master mold to easily separate the PDMS from the silicon wafer. PDMS is prepared by mixing PDMS monomer with a curing agent in a ratio of 10:1 by weight. The degassed mixture of PDMS is poured into the master mold. Afterward, the master mold filled with PDMS mixture is degassed in a vacuum desiccator to remove air bubbles inside the mixture. Finally, the cured PDMS can be peeled from the master mold and sealed to a neat glass slide via a plasma bonding process[60], [71]–[75]. The schematic illustration is shown in Fig 2.2[75]. Generating microfluidic devices via soft lithography has several advantages such as a cost-efficient process, easy to replicate, and generation of multiple microfluidic devices of the same master mold in a short time[72], [76].



Fig 2.2 Schematic illustration of fabricating microfluidic channel using soft lithography technique. First, the degassed PDMS is poured on the silicon wafer. Second, after degassing the poured PDMS on the silicon wafer, the PDMS will be cured in the oven at 90 °C for 2 hours. Then, the cured PDMS can be peeled off and sealed to a glass slide. Reproduced from ref. 75 with permission from The Royal Society of Chemistry.

PDMS microfluidic channels provide the opportunity to obtain comprehensive information on the phoretic motion of particles under the effect of a gradient in electric potential, temperature, or solute concentration. The transparent nature of PDMS and the laminar flow in the microchannels are of great help to visualize the phoretic motion of colloidal particles and analyze their migration[59]. A schematic representation of the experimental setup to observe the phoretic motion of particles is shown in Fig 2.3 which shows the syringe pumps, connecting tubes, microscope, and microchannel[77]. Based on the field gradient, the setup can be modified to create the desired gradient in the channel.



Fig 2.3 A schematic illustration of the experimental setup to visualize the phoretic motion of colloidal particles under a microscope. This figure includes two syringe pumps to inject the solution into a microchip, connecting tubes, a microchip, and a microscope to visualize the particles' motion. Adapted with permission from ref. 77. Copyright 2006 American Chemical Society.

### 2.3 Diffusiophoresis of colloidal particles in electrolyte solutions

The concept of diffusiophoresis was first introduced in the 1940s by Derjaguin et al. for nonelectrolytes in the form of capillary osmosis. Further investigations have been discussed for electrolyte solutions by Anderson and Prieve in the 1980s[44]–[46]. In reality, the presence of ions plays an important role in various applications. Therefore, we focus on the behavior of charged particles in electrolyte solutions under the concentration gradient of solute molecules or ions. Fig 2.1 shows a schematic illustration of charged particles within the concentration gradient of NaCl electrolyte[58]. For electrolyte solutions, concentration gradients of cations and anions with different diffusivities create a local electric field, in which the charged particle will experience electrophoretic motion. In the case of a thin Debye layer (Debye length ( $\kappa^{-1}$ )  $\ll$  radius of the particle (a)) and symmetric electrolyte (z:z), the electrophoretic contribution is proportional to the diffusivity difference factor  $\beta = (D_+ - D_-)/(D_+ + D_-)$  where  $D_+$  is the diffusivity of cations and  $D_-$  is the diffusivity of anions in the solution[29], [30]. Additionally, a local gradient in the osmotic pressure near the surface of particles, resulting from locally different solute concentrations, causes the chemiphoretic migration of colloidal particles. The contribution of both the electrophoresis and chemiphoresis components affects the overall diffusiophoretic velocity ( $U_{DP}$ ) of charged non-porous particles in a symmetric electrolyte (shown in Fig 2.1) which is defined as[58]

$$\boldsymbol{U}_{DP} = D_{DP} \boldsymbol{\nabla} ln \mathcal{C} \quad (2.1)$$

Here  $C(\mathbf{x},t)$  is the solute concentration as a function of position and time and  $D_{DP}$  is the diffusiophoretic mobility of particles. The diffusiophoretic mobility in the thin double-layer approximation ( $\kappa^{-1} \ll a$ ) and symmetric electrolytes is defined as[67]

$$D_{DP} = \frac{\varepsilon k_B T}{\mu z e} \left[ \beta \zeta_p + \frac{4k_B T}{z e} ln \left( \cosh\left(\frac{z e \zeta_p}{4k_B T}\right) \right) \right]$$
(2.2)

Where  $\varepsilon$  is the electrical permittivity of the solution,  $k_B$  is the Boltzmann constant, T is the absolute temperature,  $\mu$  is the viscosity of the solution, z is the valence of the ions, e is the elementary charge, and  $\zeta_p$  is the zeta potential of particles[67]. The zeta potential is the electric potential at the location of the slipping plane in the interfacial electric double layer[78]. The first term in eq (2.2) originates from the electrophoretic effect, and the second term accounts for the chemiphoresis contribution[79].

Also, the generated electric field due to the presence of ionic concentration gradient can be quantitatively calculated as[58], [80]

$$E = \frac{kT}{ze}\beta\frac{\nabla C}{C} \qquad (2.3)$$

Thus, the key components causing diffusiophoretic transport of particles are 1) the presence of a concentration gradient of solute molecule in the solution, 2) the zeta potential of the particle, and 3) a difference in diffusivities of cations and anions.

As shown in Fig 2.1, for the NaCl electrolyte solution,  $\beta$  is negative because the diffusivity of anions (Cl<sup>-</sup>) is larger than cations (Na<sup>+</sup>). In this case, a negatively charged particle tends to move toward the higher ionic concentration as a result of the electrophoretic contribution. Also, the tangential pressure gradient along the particle's surface drives the particle toward a higher concentration of ions due to the van der Waals attraction forces. Therefore, both the electrophoretic

and chemiphoretic contributions lead to the negatively charged particle's transport in the same direction in the NaCl solution[58], [81].

On the other hand, when the diffusivity of cations is higher than anions in the solution, the negatively charged particle moves toward lower ionic concentration based on the electrophoretic velocity, and the chemiphoretic velocity drives the particle toward higher concentration if there is an attractive interaction between the solute molecule and the particle. In this case, the electrophoresis and chemiphoresis will act in the opposite direction which creates a lower diffusiophoretic velocity compared to the previous case[81], [82].

For positively charged particles, the electrophoretic contribution would be different from negative particles. When the diffusivity of cations is higher than anions, the positively charged particles tend to move toward higher electrolyte concentration; whereas, positively charged particles migrate toward lower electrolyte concentration when the diffusivity of anions is larger than cations[58].

Typically, under the electrolyte concentration gradient, the particles' motion direction contributed by electrophoresis is determined by the sign of the particle's surface charge ( $\zeta_p$ ) and the relative difference in the ionic diffusivity ( $\beta$ ). If  $\beta \zeta_p > 0$ , particles migrate toward the higher electrolyte concentration, and if  $\beta \zeta_p < 0$ , particles transport along the lower electrolyte concentration[39].

#### **2.3.1** Diffusiophoresis in a microfluidic device

Verifying the enhancement of particle mobility via using microfluidics was studied in 2008 by Abécassis et al. after 60 years of development in theoretical descriptions of diffusiophoresis [69], [83]. The main physical concepts and theoretical models of diffusiophoresis can be confirmed using microfluidic channels in experiments[84]. Along with the improvement of microfluidics, more experimental studies of diffusiophoresis were reported for various channel configurations such as dead-end pores, Hele-Shaw cells, and reacting walls[29]–[31], [34], [35], [55], [67], [89].

The fundamental role of microfluidic experiments is the ability to generate the desired concentration gradient of solute molecules which leads to the corresponding particle motion. In microfluidic channels, often the observed particle migration is not solely affected by

diffusiophoresis due to the charged surfaces of the system (e.g. PDMS channel walls) and the existence of bulk fluid flow. Setting up an ionic concentration gradient naturally produces diffusioosmosis along the channel wall due to the charged surface of typical wall materials like glass, PDMS, or PMMA (poly(methyl methacrylate))[69], [78], [81], [90]. These materials have similar surface charges with commonly used particles like polystyrene, silica, etc., which means that the order of magnitude of the diffusioosmotic velocity along the channel wall and the particles' diffusiophoretic velocity are comparable. The influence of diffusioosmosis is one of the crucial complications in microfluidic analyses for calculating the experimental diffsuiophoretic velocity. Therefore, the effect of the diffusioosmotic slip velocity should be considered in calculating the net diffusiophoretic velocity in the microfluidic channel[30], [67].

#### 2.3.2 Diffusiophoresis in a dead-end pore channel

Manipulation of colloids in dead-end geometries is commonly used in various applications such as drug delivery and underground oil and gas recovery. In this kind of geometry, Brownian motion is the sole natural mechanism that enables the transport of colloids into or out of pores, but this mechanism is extremely slow and inefficient for microscale particle migration. Thus, there is a scope for increasing the transport of colloids into or out of the pore through introducing a solute gradient. Depending on the ions' diffusivity, the migration of colloidal particles into the dead-end pore can be either enhanced or prevented[55], [85].



Fig 2.4 Sketch of a set of dead-end pores in the microfluidic channel. The upper section is the main channel and the small branches are called dead-end pores. Adapted with permission from ref. 29. Copyright 2020 American Chemical Society.

A dead-end pore geometry allows for generating a one-dimensional concentration gradient of solutes to analyze and visualize the diffusiophoretic motion of colloids. A schematic representation of a series of dead-end pores in a microfluidic device is shown in Fig 2.4[29]. The upper side in this figure is the main channel where the fluid flows in it via connecting a syringe pump to the entrance of the channel. The branches of the channel are dead-end pores where the solute concentration gradient is established along the length of the pores.

In contradiction to electrophoresis or thermophoresis where the gradient field is typically constant, the produced concentration gradients in this configuration are time-dependent. To generate a concentration gradient of solute in this channel, the entrance of the main channel is connected to a syringe pump that contains solution 1. Once solution 1 has filled the main channel, blocking the outlet of the channel drives the solution into the pores. After filling the whole channel with solution 1, the plug blocking the outlet is removed and the main channel is evacuated using an air plug. Once the main channel is empty, solution 2 will be injected into the channel. To prevent the mixing of two solutions, an air plug is introduced into the main channel before injecting solution 2[29], [30], [67].

Typically, two different approaches have been followed in this geometry to create the solute concentration gradient and visualize the particles' migration. The first approach is a compaction experiment where the pores are initially filled with solution 1 which is the colloidal suspension. Then, solution 2 which is the solution without particles is entered into the channel to produce the concentration gradient as shown in Fig 2.5[29].



Fig 2.5 Generating concentration gradient through the compaction experiment in a dead-pore channel. Initially, the pores are filled with the colloidal suspension (solution 1). Then an air plug following with the second solution is injected into the channel. The starting time of the experiment is after the two solutions come into contact at the entrance of the first pore. Adapted with permission from ref. 29. Copyright 2020 American Chemical Society.

The second technique for creating a concentration gradient in dead-end pores channels is an entrainment experiment where the pores are filled with solution 1 which is the electrolyte solution without colloids. Afterward, the colloidal suspension (solution 2) is injected into the channel after evacuating the main channel with the air bubble which is shown in Fig 2.6.



Fig 2.6 Creating concentration gradient using entrainment approach in a dead-end pore microfluidic channel. First, the pores are filled with the electrolyte solution (solution 1), then, the colloidal suspension (solution 2) is flown into the channel after evacuating the main channel. Adapted with permission from ref. 29. Copyright 2020 American Chemical Society.

The critical issue in the compaction experiment is the concentration of the electrolyte in the colloidal suspension because the particles stuck to the channel walls or aggregate when suspended in the solution with a high concentration of electrolyte (such as 10mM). For this study, we select the entrainment experiment to analyze and observe the diffusiophoretic motion of colloids to avoid the aggregation and sticking of colloids to the channel walls problems.

Furthermore, various studies verify that particle capture and migration along the pore length will not happen in the absence of a concentration gradient in the pores. As shown in Fig 2.7(a), the particles can penetrate the entrance of the pore via hydrodynamic forces and the presence of vortexes but they cannot diffuse more along the length of the pores and ultimately escape[67]. The penetration length of particles at the entrance of the pores due to the hydrodynamic forces is called entrance length. There are various equations to calculate the entrance length depending on the channel geometry and the range of Reynolds numbers[91]–[95]. For rectangular geometry which is common geometry used for the diffusiophoresis phenomenon, the entrance length can be calculated as[94]

$$L_e = (w_{cs}) * \left(\frac{0.55}{1 + 0.13Re} + 0.065Re\right)$$
(2.4)

 $L_e$  is the entrance length, w<sub>cs</sub> is the width of the cross-section and Re is the Reynolds number which is defined as Re=puw<sub>cs</sub>/µ for a rectangular channel in which w<sub>cs</sub> is the larger side.



Fig 2.7 The trajectories of 1µm particles in a series of dead-end pores using the entrainment experiment approach with and without the presence of a concentration gradient. (a) In the absence of a concentration gradient in the pores, colloids will not be captured inside the pores and they just enter the entrance of the pores due to the hydrodynamic forces and flow geometry effect. (b) In the presence of an electrolyte concentration gradient (10mM of NaCl inside the pore and 0.1mM outside the pore), the particles migrate into the pores as a result of the diffusiophoretic velocity. The colloid tracks in the main channel are fainter than (a) due to the difference in the contrast settings during the image processing. Reproduced from ref. 67 with permission from The Royal Society of Chemistry.

On the other hand, the particles diffuse inside the pores in the presence of a concentration gradient shown in Fig 2.7(b)[67]. In the presence of a concentration gradient in dead-end pores, the velocity of colloidal particles at the entrance of the pores is a balance between the convective velocity  $(\mathbf{U}_{conv})$  and the diffusiophoretic velocity  $(\mathbf{U}_{DP})$  as displayed in Fig 2.8[51]. Therefore, the convective velocity, which is controlled by the pump flow rate, plays an important role in the dead-end pore geometry. As the flow rate changes, the balance between the diffusiophoretic velocity and the convective velocity changes as well. For instance, increasing the flow rate will increase the influence of the convective velocity on the net velocity and make it to be more dominant which means the possibility of particles' capturing inside the pores will diminish.



Fig 2.8 Schematic illustration of the imposed velocities on the colloidal particle in the presence of a concentration gradient at the entrance of dead-end pores. The convective velocity originated from the ambient flow of the particle, the diffusiophoretic velocity is created due to the imposed concentration gradient in the pores, and the net velocity is the final velocity of the particle based on the balance between the diffusiophoretic velocity and convective velocity.

#### 2.3.3 PDMS interaction with the dissolved CO<sub>2</sub> in a solution in the microfluidic channel

When a suspension of colloidal particles flows in a microfluidic channel at a constant flow rate, a uniform dispersion of particles in the channel is expected. However, an unexpected accumulation of colloidal particles can take place in the channel without imposing any gradient inside the microfluidic channel[96]. The spontaneous transport of particles can happen due to gas leakage through permeable walls and the interaction of the PDMS wall with a solution. PDMS is the most popular material to fabricate microfluidic channels which has a high gas permeability and a negative surface charge [96]. The dissolution of the atmospheric  $CO_2$  in the aqueous solution and the interaction of the pre-dissolved CO<sub>2</sub> with the PDMS wall can have a large impact on the microscale transports[97]. Dissolution of CO<sub>2</sub> in an aqueous solution (CO<sub>2</sub> + H<sub>2</sub>O  $\leftrightarrow$  H<sup>+</sup> + HCO<sub>3</sub>,  $HCO_3^- \leftrightarrow H^+ + CO_3^{2-}$  generates  $H^+$ ,  $HCO_3^-$ , and  $CO_3^{2-}$  ions. This dissolution of gas can create a self-generated ions concentration gradient in the flow which can drive particles in a specific direction due to the diffusiophoresis phenomenon[96]. Also, the importance of this small amount of concentration gradient stems from the large difference in the H<sup>+</sup> diffusivity ( $D_{H^+}$  =  $9.3 \times 10^{-9} m^2 s^{-1}$ ) and HCO<sub>3</sub> diffusivity ( $D_{HCO_3} = 1.2 \times 10^{-9} m^2 s^{-1}$ ). This large difference creates a large electric field based on eq (2.3) resulting in the transport of colloidal particles in the channel.

Therefore, a self-generated concentration gradient of solute can form in a microfluidic channel and create a diffusiophoretic motion of charged colloids. This spontaneous phenomenon can suggest various potential applications such as particle collecting and concentrating. In the first part of our

work, we investigate the influence of the dissolved  $CO_2$  in DI water and its interaction with the PDMS walls in a dead-end pore geometry on the particles' motion.

# This wants in a dead end pore geometry on the particles' motion.

**2.4** Effect of the interior structure of colloids on their diffusiophoretic motion The diffusiophoretic motion of non-porous particles has been widely explored in various theoretical and experimental studies [27]–[30], [57], [98]–[101]. However, many colloidal particles have a porous structure and they are permeable to the liquid medium such as proteins, DNAs, porous silica, etc[102]–[104]. Thus, understanding the diffusiophoretic motion of porous particles both theoretically and experimentally is very important for applications in drug delivery, wastewater, and separation fields where using porous particles is relevant[105]. There are two crucial differences between non-porous particles and porous particles which distinguish their diffusiophoretic motion. The first difference is the overall hydrodynamic drag force for porous particles is much smaller than for hard particles due to the permeability of fluid through the interior region of porous particles[38]. The second difference arises from the distribution of charges over the entire interior of porous particles; however, the surface charges of hard particles are distributed only on the particle surface. In the case of the thin double layer ( $\kappa^{-1} \ll a$ ), the influence of the reduced hydrodynamic drag force and distribution of surface charges all over the interior region of porous particles can cause orders of magnitude larger mobility compared to hard particles [37], [38], [105]–[107].

Moreover, the fixed charges carried by the porous particle can be neutralized by driving counterions into the interior region of porous particles when the Debye length layer is thin (large value of ka). This phenomenon is called "counterion condensation" or "shielding effect" which can reduce the chemiphoretic contribution to the phoretic motion of porous particles[38], [105], and even can neutralize the effect of chemiphoresis on the particle's mobility because the porous particle at very large ka is essentially chargeless.

The diffusiophoretic motion of porous particles has been investigated in a few theoretical studies in different aspects but there is a gap in the experimental analysis of the motion of the porous particles under the concentration gradient[38]–[41], [43], [69]. Thus, we experimentally investigate the difference in porous and hard particles' migration in the presence of a solute concentration gradient.

Eq (2.1) & (2.2) are derived for calculating the diffusiophoretic velocity of hard sphere particles. To drive the diffusiophoretic velocity for porous particles, the modified Stokes equation for Newtonian fluid with an extra term accounting for the local electric body force should be considered as follows [43]

$$-\nabla p + \mu \nabla^2 \mathbf{v} - \rho \nabla \phi - \gamma \mathbf{v} = 0, \quad 0 < r < a$$
(2.5)  
$$-\nabla p + \mu \nabla^2 \mathbf{v} - \rho \nabla \phi = 0, \quad a < r < \infty$$
(2.6)

 $\rho$  is the space charge density of mobile ions in the system,  $\gamma$  is the friction coefficient representing the reduced drag force in a porous material, p is the fluid pressure, **v** is the fluid flow, and  $\phi$  is the electric potential. The permeability of the porous particles can be characterized by using the Brinkman screening length ( $\lambda^{-1}$ ). The Brinkman screening length is calculated through  $\lambda^{-1} = (\gamma/\mu)^{-1}$ <sup>0.5</sup>, and  $\lambda a$  is its dimensionless form. As the  $\lambda a$  becomes smaller, the particle permeability of the fluid is increased[41], [43]. The first term ( $-\nabla p$ ) in eq (2.5) states the pressure gradient which prevents motion due to normal stresses. The second term ( $\mu \nabla^2 \mathbf{v}$ ) represents the internal stress forces or viscous effects. The third term and fourth terms show the local electric body force and the reduced drag force inside porous particles, respectively[108].

Considering the modified Stokes equation, Gauss theorem, the conservation of ions, and the incompressibility constraint of the fluid flow resulting in the following diffusiophoretic velocity for porous particles[43]

$$\boldsymbol{U}_{DP} = \frac{\varepsilon}{\mu} \frac{|\nabla C|}{C_0} \left(\frac{kT}{ze}\right)^2 \left[ (\kappa a)^2 \beta A_1 \bar{Q} + (\kappa a)^4 A_2 \bar{Q}^2 \right] + O(\bar{Q}^3) \quad (2.7)$$

 $C_0$  is the initial electrolyte concentration, a is the radius of the colloidal particle,  $\bar{Q} = \frac{ZeQ}{\varepsilon \kappa^2 kT}$  is a dimensionless fixed-charge density,  $A_1$  and  $A_2$  are functions of permeability, the volume fraction of particles, and the electrokinetic radius ( $\kappa a$ ).  $A_1$  represents the electrophoretic contribution in the diffusiophoretic velocity, and  $A_2$  shows the contribution of chemiphoresis.



Fig 2.9 The dimensionless diffusiophoretic velocity for charged porous particles as a function of dimensionless fixed charge density in the suspension with volume fraction 0.1 and  $\lambda a = 1$  ( highly permeable particles). (a) The dimensionless diffusiophoretic velocity when  $\beta = 0$  means only chemiphoresis is contributing to trigger diffusiophoretic velocity. (b) The dimensionless diffusiophoretic velocity when  $\beta = -0.2$  means both electrophoresis and chemiphoresis are contributing to the diffusiophoretic velocity. Adapted with permission from ref. 43. Copyright 2015 American Chemical Society.

The variation of  $A_1$  and  $A_2$  functions have been studied by Huang et al.[43] for different ranges of permeability, volume fraction, and Debye length layer. They have shown that the diffusiophoretic mobility in a suspension of porous particles is a complex function of the permeability of the colloidal particle, volume fraction, surface charge of particles, the relative difference in the anions and cations diffusivities ( $\beta$ ), and Debye length in the solution. Also, Lee's theoretical study[38] verifies the complex dependency of porous particles' diffusiophoretic velocity on the mentioned parameters; whereas, the diffusiophoretic mobility of hard spheres (eq (2.2)) depends on the zeta potential of particles and the relative difference in the ions' diffusivity.

Furthermore, two important ranges of  $\kappa a$  can completely change the behavior of porous particles under a concentration gradient. For finite  $\kappa a$  (less than 1) which means the size of the Debye layer is comparable with the particle's radius, the double-layer polarization effect becomes important, and for large  $\kappa a$ , the counterion condensation effect should be considered. As shown in Fig 2.9(a), the diffusiophoretic velocity of charged porous particles with increasing  $\kappa a$  decreases. For large  $\kappa a$ , it shows the diffusiophoretic velocity which is due to the counterion condensation effect. Also, the diffusiophoretic velocity of porous particles as a function of dimensionless fixed charged density in different  $\kappa a$  in the presence of both chemiphoretic and electrophoretic contributions is shown in Fig 2.9(b). The diffusiophoretic velocity in this picture is more complicated than part (a) where just chemiphoretic contribution is available. As shown in Fig 2.9(b), the diffusiophoretic velocity increases as  $\kappa a$  increases but for the small  $\kappa a$ , the behavior is not monotonic due to the double-layer polarization effect. Moreover, in the range of low dimensionless fixed charge density (( $\kappa a$ )<sup>2</sup> $\bar{Q} < 1$ ), the dimensionless diffusiophoretic velocity is very small and close to zero for all values of  $\kappa a$ . Therefore, both the dimensionless fixed charge density and  $\kappa a$  values are critical parameters in calculating the diffusiophoretic velocity.

In our work, we experimentally studied the effect of the porosity of particles on their diffusiophoretic motion in a microfluidic channel for the case of large  $\kappa a$  and compared the results for porous particles with hard particles under a similar electrolyte concentration gradient.

Therefore, two topics in the diffusiophoretic in a dead-end pore microfluidic channel will be investigated in this thesis. The first one is to study the effect of dissolved  $CO_2$  and the impact of PDMS surface chemistry in altering its equilibrium dissociation in DI water on the particles' motion in the dead-end pore microfluidic channel without introducing any concentration gradient to the channel. The second one is to explore the diffusiophoretic motion for porous particles as a function of pore size (which changes the porosity as well) and compare it to non-porous particles under the same conditions.

## **Chapter 3**

# 3 A self-generated exclusion zone in a dead-end pore microfluidic channel

The work presented in this chapter is currently under review for publication as a research article in Lab On a Chip Journal.

#### 3.1 Introduction

One of the most striking self-driven phenomena is the natural migration of colloidal particles under a self-generated or imposed concentration gradient, called diffusiophoresis. This migration of colloidal particles finds applications in water filtration[68], [109]-[111], microfluidic microdialysis[112], antifouling coatings[54], [89], [113], actuators and self-propelled motors[114], [115]. The repulsion of colloidal particles in an aqueous solution from an interface over distances of up to hundreds of micrometers results in a particle-free zone in the vicinity of the surface even if there are no external driving forces present. This so-called exclusion zone phenomenon has been observed near a wide range of materials, including biological tissues, hydrogels, hydrophilic polymers, and metals [116]-[119]. One of the intriguing characteristics of the exclusion zone is the remarkably long particle-repelling range which extends far beyond the Debye length [88], [89], [120]. Generally, the thickness of the exclusion zone varies from several tens to several hundreds of micrometers depending on the suspended colloidal particles, properties of the surface, the medium, and pH [116], [121]. While various explanations and mechanisms have been proposed for the exclusion zone formation, the exact physical origin has remained ambiguous [119], [122]-[125]. However, recent studies have suggested that a combination of a pH gradient near the surface, ion-exchange at the surface, diffusion of ions, and diffusiophoresis of suspended particles in the resulting ions' concentration gradient are playing key roles in the formation of the exclusion zone [31], [88], [110], [119], [123]. The concentration gradient of ions causes the migration of colloidal particles because of electrophoresis and chemiphoresis. The concentration gradient of ions with different diffusivity produces a local electric field triggering colloidal particles' motion in the solution through electrophoresis. Simultaneously, the

concentration gradient of ions creates an osmotic pressure gradient along the particle's surface, which contributes to the colloidal particles' motion through chemiphoresis resulting in particle propulsion towards the higher or lower concentration of ions [57], [58]. The direction of particles' motion in the concentration gradient depends on the surface charge of the particles, the sign and magnitude of the local electric field, and the interaction of particles with solute molecules which can be attractive or repulsive.

Most recent studies have used ion-exchange membranes such as Nafion in a microfluidic channel to generate the exclusion zone [31], [52], [88], [110], [119], [126]. Ion-exchange membranes create a gradient of ions, which in turn causes the colloidal particles to repel from the membrane surface and form the exclusion zone in the channel. In these systems, a combination of larger cations (typically Na<sup>+</sup> or K<sup>+</sup>) and smaller cations (H<sup>+</sup>) is present, complicating the description of local ion and charge distributions.

In this work, we utilize a dead-end pore geometry and we benefit from the interfacial chemistry of the microfluidic device to create a self-generated concentration gradient of ions. In the past decade, a dead-end pore geometry was the prevalent configuration to generate concentration gradients in microfluidic chips to investigate diffusiophoretic transport at the microscale [29], [67], [69]. Although the concentration gradient generating procedure is entirely different in our system from Florea's work[31], physical principles in the formation of the exclusion zone are similar to their study. However, the only cation present is H<sup>+</sup> in our system, simplifying its physical understanding. We do not use any salt or other solute to generate the ionic concentration gradient, which is simply developed by the interfacial interaction between the PDMS wall and DI water.

We combine experimental observations and analytical investigation of the exclusion zone formation. A boundary layer model was developed to analytically calculate the exclusion zone thickness. We experimentally evaluated the exclusion zone thickness as a function of the particles' positions. Furthermore, we explore the transient dynamics of the exclusion zone in our system for two different sizes of silica particles. Our theoretical analysis is in good agreement with the experimental results demonstrating that the exclusion zone thickness is highly dependent on the Sherwood number.

## 3.2 Experimental setup

#### 3.2.1 PDMS Microfluidic Channel

We perform experiments on a microfluidic chip with a design and dimensions as shown in Fig 3.1. We used a dead-end pore geometry that includes two sections, a main channel, and side channels which are referred to as pores.

The width and height of the main channel, respectively, are 300  $\mu$ m and 90  $\mu$ m. The height and length of the dead-end pores are 30  $\mu$ m and 1500  $\mu$ m, respectively. In our design, we considered 3 different values for the width of the pores which are 50  $\mu$ m ( $w_{p1}$ ), 100  $\mu$ m ( $w_{p2}$ ), and 150  $\mu$ m ( $w_{p3}$ ). The height of dead-end pores is thinner than the main channel to reduce the disturbance from the main channel inside pores[85]. Additionally, the narrow pore geometry (width << length) allows for neglecting the pressure gradient in the y-direction[30], and considering a one-dimensional motion of colloidal particles in the x-direction. Therefore, this channel design provides a good condition to analyze the motion of colloidal particles under the influence of a transient one-dimensional concentration gradient[127].



Fig 3.1 Schematic configuration of the dead-end pore geometry and dimensions. After filling pores with DI water, a suspension of silica particles flows through the evacuated main channel and reaches a series of dead-end pores containing DI water.

The PDMS channels are prepared by standard soft lithography procedures where the mixing ratio of the elastomer base (PDMS) and the curing agent (Dow Corning, Sylgard<sup>®</sup> 184) was 10:1. Subsequently, the mixture goes through a desiccation process in the vacuum desiccator to remove air bubbles. Then, we pour the mixture on the patterned master mold and heat it to 90 °C for 2 hours after the desiccation process. In the final step, we peel back the cast PDMS and plasma-bond it to a glass slide. We use the prepared channel for running every experiment after 24h.

#### **3.2.2 Experimental Protocol**

We prepared two suspensions of silica particles with different diameters, 1  $\mu$ m, and 2  $\mu$ m (SOLAD<sup>TM</sup> particle, Glantreo) suspended in DI water at the volume fraction of 2.6 × 10<sup>-4</sup> to ensure that we are in the dilute concentration. We use a bath sonicator for 20 min to suspend silica particles in DI water. We initially inject pure DI water without particles into the chip using a syringe pump (Harvard Apparatus). Once both the main channel and pores were filled, we introduced an air bubble into the main channel which is followed by the second solution containing colloidal suspension in DI water with a flow rate of 250 µL/h. When the colloidal particle suspension reaches the entrance of the first pore, the mean flow rate is reduced to 30 µL/h.

We start recording videos as soon as the colloidal suspension (second solution) reaches the entrance of the first pore. Observations were made at 10× magnification for most cases using the ECHO microscope (RVL-100-G, San Diego CA, USA). General image processing is carried out in ImageJ.

#### **3.3 Results and discussion**

#### **3.3.1** Exclusion Zone in the dead-end pore geometry

In our setup, charged silica particles ( $\zeta_p = -37 \text{ mV}$ ) are suspended in DI water and based on the chemical equilibrium of water ( $H_2 O \rightleftharpoons H^+ + OH^-$ ) and dissolution of carbon dioxide in a solution exposed to the atmosphere the available ions in the solution are  $H^+$ ,  $OH^-$ ,  $HCO_3^-$ , and  $CO_3^{2-}$ . The concentration of these ions in the pores is different from their concentration in the colloidal suspension due to the ions' interfacial interactions with channel walls in the pores which have a negative surface charge ( $\zeta_{PDMS} = -49 \text{ mV}$  [128]).


Fig 3.2 A schematic representation (a) and experimental observation (b) of the exclusion zone formation in the main channel caused by a self-generated concentration gradient from the dead-end pores. (a) The blue dashed line represents the ionic exclusion zone, which is larger than the particle exclusion zone represented by the red line. A particle in the main channel experiences forces in the tangential direction as a result of the ion concentration gradient. Experimental observation of the exclusion layer formation in the dead-end channel for pore width =  $100 \mu m$ .

Therefore, as colloidal particles pass through the channel and reach the first pore, the concentration gradient of ions starts to develop through the pores into the main channel. As a result, the suspended particles tend to migrate away from the pores of the channel over time as shown in Fig 3.2 ( also see Appendix A), and a temporary arc-shaped exclusion zone is developed along the y-direction which is caused by ion diffusion and ion concentration gradient, together with a constant flow of colloidal suspension.

The formation of the exclusion zone by colloidal particles depends on the balance between diffusiophoretic velocity and convective velocity as displayed in Fig 3.2. The diffusiophoretic velocity of a particle ( $\mathbf{U}_{DP}$ ) is proportional to the gradient of the logarithm of the ion concentration as [67]

$$\boldsymbol{U}_{DP} = D_{DP} \boldsymbol{\nabla} ln \mathcal{C} \quad (3.1)$$

Where  $C(\mathbf{x},t)$  is the solute concentration as a function of position and time and  $D_{DP}$  is the diffusiophoretic mobility of particles which in the case of a thin Debye layer (Debye length  $\ll$  radius of the particle) and symmetric electrolytes is defined as [67]

$$D_{DP} = \frac{\varepsilon k_B T}{\mu z e} \left[ \frac{D_+ - D_-}{D_+ + D_-} \zeta_p + \frac{4k_B T}{z e} ln \left( \cosh\left(\frac{z e \zeta_p}{4k_B T}\right) \right) \right] \quad (3.2)$$

Where  $\varepsilon$  is the electrical permittivity of the solution,  $k_B$  is the Boltzmann constant, T is the absolute temperature,  $\mu$  is the viscosity of the solution, z is the valence of the ions, e is the elementary charge,  $\zeta_p$  is the zeta potential of particles,  $D_+$  is the cation diffusivity, and  $D_-$  is the anion diffusivity. Furthermore, based on eq (3.1), the diffusiophoretic velocity depends on  $\nabla \ln C$ , implying that only relative changes in the solute concentration affect the diffusiophoretic velocity, and even at extremely low solute concentrations, the diffusiophoresis phenomenon can cause the formation of exclusion thickness[31]. The convective velocity can be adjusted by changing the inlet flow rate to the main channel. In the following, theoretical analysis and experimental results of the formation of the exclusion zone are discussed.

#### 3.3.2 Ion's boundary layer analysis

The available ions present in our system are  $H^+$ ,  $OH^-$ ,  $HCO_3^-$ , and  $CO_3^{2-}$ , as a result of the interactions of DI water with the PDMS walls. When multiple ions are present in the system, predicting the ion concentration profile analytically by using the diffusion-advection equation is difficult due to coupling in ion concentrations. Four Nernst-Planck equations should be written for each of these ions separately. If the local electroneutrality, the chemical equilibrium of water, and the second chemical equilibrium of  $CO_2$  dissolution ( $HCO_3^- \rightleftharpoons H^+ + CO_3^{2-}$ ) are satisfied, these four equations can be reduced to one equation for the ionic pair  $H^+ - HCO_3^-$ ,. These ions can

create a large diffusion potential due to the large difference in their diffusivities ( $D_{H^+} = 9.3 \times 10^{-9} m^2 s^{-1}$  and  $D_{HCO_3^-} = 1.2 \times 10^{-9} m^2 s^{-1}$ ) based on the first term in eq (3.2)[111]. The simplified Nernst-Planck equation relates the concentration of the  $H^+$  to its diffusion flux and the background flow velocity as [31], [110], [129]

$$\frac{\partial c_H}{\partial t} = \nabla \cdot \left( D_{Heff} \nabla c_H - u c_H \right) \quad (3.3) \qquad D_{Heff} = \frac{2 D_H D_{HCO_3}}{D_H + D_{HCO_3}} \quad (3.4)$$

Where **u** is the background flow velocity,  $c_H$  is the concentration of  $H^+$  and  $D_{Heff}$  is the effective diffusivity that can be obtained through eq (3.4). The governing boundary conditions in our system are:

$$c_{H} = c_{H\infty} \text{ at } y = 0 \text{ (inlet)}$$

$$n. (-D_{H} \nabla c_{H} + uc_{H}) = n. (uc_{H}) \text{ at } y = L_{ch} \text{ (outlet)}$$

$$\frac{dc_{H}}{dx} = 0 \text{ at } x = \frac{W_{ch}}{2}$$

$$(3.7)$$

Where  $c_{H\infty}$  is the bulk  $H^+$  concentration in the main channel, **n** is the outward normal vector, and  $L_{ch}$  is the channel length. Applying the 'boundary layer theory' as shown in Appendix A, the exclusion zone thickness can be obtained as follows

$$\delta = \sqrt{\frac{8}{3} \frac{D_{Heff}}{\bar{u}_y} y} \qquad (3.8) \quad \to \quad \delta = w_{ch} \sqrt{\frac{8}{3} \frac{1}{Sh} \frac{y}{L}} \qquad (3.9)$$

Where Sh is the Sherwood number representing the ratio of longitudinal convective transfer to the transverse diffusion rate which links most of the physicochemical parameters (flow rate, diffusivity of ions, and geometrical factors) of the system.

$$Sh = \frac{w_{ch}^2 \bar{u}_y}{D_{Heff}L} \quad (3.10)$$

Hence the only non-geometrical parameter that controls the problem and thus, Sh, is the mean flow velocity. As the mean flow velocity increases, the migration of ions is dominated by the convective velocity and the boundary layer of ions perpendicular to the flow direction is smaller causing a smaller exclusion zone[130].

#### 3.3.3 Particles' trajectory analysis

As explained in the previous section, colloidal particles are moving in the main channel based on the balance between diffusiophoretic force, convective drag, and Brownian force. In our system, the particles' velocity in the x-direction is dominated by diffusiophoretic velocity, and the particles' velocity in the y-direction is dominated by convective velocity. In these assumptions, the exclusion zone for colloidal particles based on their trajectory is the following formula in the limit of steady state (see SI section A3 for more detail)

$$\delta_p = \frac{D_{DP}}{C_{H\infty}} C_{H0} \sqrt{\frac{2}{3} \frac{1}{D_{Heff} \bar{u}_y} y}$$
(3.11)

Eq (3.11) is a function of the bulk ion concentration in the pore and the bulk ion concentration in the main channel which cannot directly be measured experimentally. Fig 3.3 shows the experimental results multiplied by the square root of velocity for two different flow rates presenting the same results and verifying that eq (3.11) can accurately describe the main relevant physical mechanisms in our experiments. Also, the ratio of normalized particle exclusion thickness at two different flow rates is shown in figure A2 in Appendix A which shows a good agreement between theoretical and experimental results.



Fig 3.3 The experimental data of normalized particle exclusion thickness multiple by the square root of velocity for 2  $\mu$ m silica particles. The orange square shows experimental data in Q=80  $\mu$ L/hr, and the black star is for 30  $\mu$ L/hr. The pore width in this diagram is 100  $\mu$ m.

Besides, a comparison of experimental data with theoretical ions' migration for exclusion zone thickness is shown in Fig 3.4 using eq (3.9). For different pore widths  $(w_p)$ , the theoretical predictions are qualitatively similar to experimental observations. The particle exclusion zone is slightly smaller than the predicted ionic gradient zone, which is in line with expectations.



Fig 3.4 Comparing the exclusion zone thickness as a function of position over the first pore calculated in theory with experimental results for 2  $\mu$ m silica particles. (a) For pore width=50  $\mu$ m. (b) For pore width=100  $\mu$ m. (c) For pore width=150  $\mu$ m. The red curve represents the ionic exclusion thickness and the black curve displays the silica particles exclusion thickness.

Using values of the physical parameters in our system to determine the diffusiophoretic mobility, we find  $D_{DP} = 6.19 \times 10^{-10} m^2 s^{-1}$  (eq (3.2)), which is orders of magnitude larger than the expected diffusion coefficient for 2µm silica particles in water calculated using the Stokes-Einstein equation;  $D_P = \frac{k_B T}{6\pi\mu r} = 2.18 \times 10^{-13} m^2 s^{-1}$ . On the other hand, the migration of ions in a concentration profile depends on the effective diffusivity which is equal to  $2.094 \times 10^{-9} m^2 s^{-1}$  (based on eq (3.4)) in our system. Thus, the effective diffusivity for ions is larger than the particles' diffusiophoretic mobility which gives rise to generating the larger ion exclusion thickness in theory compared to the experimental results for particles as presented in Fig 3.4.

It is worth mentioning that a substantial factor that can affect the diffusiophoretic mobility value is the zeta potential of colloidal particles as shown in eq (3.2). Lee et al. [110] have presented that when the magnitude of the zeta potential of colloidal particles ( $\zeta_p$ ) is larger, suspended particles will be excluded further away from the ion-exchange membrane and tend to form a larger exclusion zone in the channel because the diffusiophoretic velocity is stronger to initiate the formation of an exclusion zone. In our system, the zeta potential of silica particles was -37 mV approximately which confirms the difference between the particles' exclusion zone and boundary layer analysis prediction shown in Fig 3.4 based on Lee's results [110].

The thickness of the exclusion zone presented in Fig 3.4 is comparable with previous studies that utilized salt and ion exchange membrane to create an exclusion zone [110]. While in our setup, we do not use any salt or membrane to generate the concentration gradient. This is solely generated due to the geometry of the channel and the interfacial interaction of species with the channel wall, indicating that these interactions can have a significant influence on colloidal transport. In our experiments, the development of the exclusion zone in the main channel stops after the self-generated ion concentration gradient disappears (t~90 s) indicating that at initial experimental times, this interfacial effect is non-negligible.

#### **3.3.4** Evolution of the Exclusion zone

An advantage of the dead-end pore geometry is the possibility to study the kinetics behind the exclusion zone formation. The exclusion zone shown in Fig 3.2 diminishes after the ion concentration gradient fades due to the balance between the pores and the main channel. This means that the formation of the exclusion zone in the main channel is temporary because pores contain a finite amount of ionic species through the wall-fluid interactions which create this temporary concentration gradient in the channel. Measurements of the exclusion zone thickness as a function of time are presented in Fig 3.5 and Fig 3.6 for 2µm and 1µm silica particles, respectively. The exclusion zone thickness shown in Fig 3.5 and Fig 3.6 was normalized by the diffusion displacement. As the concentration gradient reduces over time, the exclusion zone thickness shrinks as well because the magnitude of diffusiophoretic velocity decreases.



Fig 3.5 Normalized exclusion zone thickness over time for  $2\mu m$  silica particles. (a) For pore width=50 $\mu m$ . (b) For pore width=100 $\mu m$ . (c) For pore width=150 $\mu m$ .



Fig 3.6 Normalized exclusion zone thickness over time for  $1\mu m$  silica particles. (a) For pore width=100 $\mu m$ . (b) For pore width=150 $\mu m$ .

Both Fig 3.5 and Fig 3.6 show that the ratio of exclusion zone thickness to the diffusion displacement diminishes over time which means that the movement of silica particles in the x-direction is getting dominated by Brownian motion gradually due to the dwindling of ion concentration profile. In Fig 3.5(c), an increase in the exclusion thickness in the first few moments (t=1 min and t=2.5 min) and decreasing after reaching the maximum exclusion thickness (at t=2.5 min) can be observed. Although the exclusion zone thickness created in our system is comparable

with previous studies, the stability of the generated exclusion zone is limited because the ionic concentration profile is temporary and there is no renewal of the gradient in our system.

Moreover, the results for different pore widths shown in Fig 3.5 and Fig 3.6 indicate that the width of pores does not influence the transient evolution of the exclusion zone significantly because the number density of ions available in the pores does not change. The number density of ions is the key parameter controlling the stability of the exclusion zone which is constant for all three widths of pores. Thus, altering the width of pores between 50  $\mu$ m and 150  $\mu$ m does not affect the stability of the concentration gradient in the main channel which is in line with expectation as the number density of ions is constant in the pores.



Fig 3.7 Effect of the particle size on the sustainability of the exclusion zone over time. This figure is plotted for the end-point ( $y = 150\mu m$ ) of the first pore. The red curve and black curve represent  $2\mu m$  and  $1\mu m$  particles, respectively.

The exclusion zone's transient is different for 2 µm and 1 µm silica particles as shown in Fig 3.7. Both sizes of silica particles need almost the same time to reach the maximum exclusion zone thickness (~140 s); however, after reaching the maximum point, reduction in the exclusion zone thickness happens faster for 1 µm silica particles sample compared to 2 µm particles. This is because the diffusion coefficient of 1 µm particles ( $D_P = 4.36 \times 10^{-13} m^2 s^{-1}$ ) is two times higher than the one for 2  $\mu$ m particles and accelerates the dwindling of the exclusion zone. As a result, the exclusion zone over the pore diminishes in a shorter time in the case of 1  $\mu$ m silica particles (~ 600s) in comparison with 2 $\mu$ m silica particles (~ 1050s). This indicates that the size of silica particles is an important factor in the transient dynamics of the exclusion zone.

#### 3.3.5 Balance between diffusiophoretic velocity and convective velocity

As mentioned before, the establishment of the exclusion zone in the main channel depends on the net balance between diffusiophoretic velocity and convective velocity. Fig 3.8 demonstrates the influence of flow rate on exclusion zone thickness for three different injecting flow rates. A higher flow rate results in a higher Sh which triggers shrinkage in the exclusion zone thickness because it boosts the longitudinal convection. This behavior was experimentally observed as well; as we increased the flow rate from  $30 \,\mu$ L/hr to  $80 \,\mu$ L/hr, the exclusion zone thickness formed by particles reduced as shown in Fig 3.8(a).



Fig 3.8 Effect of increasing flow rate on the maximum exclusion thickness based on the theoretical calculations and experimental observations for three different flow rates. The pore width in this diagram is 100  $\mu$ m. Dashed lines show the ionic exclusion thickness and solid lines represent 2  $\mu$ m silica particles exclusion thickness. (a) maximum exclusion thickness for each flow rate. (b) Scaling the dimensionless exclusion thickness by the inverse of the square root of the Sh number.

Also, the exclusion zone thickness reduction, with increasing the flow rate, in the experimental observation is larger than the theoretical prediction due to the difference in the diffusiophoretic mobility and the effective diffusivity of ions  $(\frac{D_{DP}}{D_{Heff}} \cong 0.3)$ . As the diffusiophoretic mobility of silica particles is less than the effective diffusivity of ions, the longitudinal convective effect would be more dominant causing a larger reduction in the exclusion thickness in experimental data compared to the theoretical analysis (eq (3.9)). Moreover, Fig 3.8(b) verifies the dependency of exclusion thickness on the inverse of the square root of Sh as eq (3.9). The dimensionless exclusion thickness scales with the inverse of the square root of the Sh number and the experimental and theoretical data for different flow rates collapse. This effect is in qualitative agreement with the findings of Lee et al. for different Sh numbers [110].

#### 3.4 Conclusion

We investigated a temporary exclusion zone mechanism by a self-generated concentration gradient. In previous studies [31], [88], [110], an ion exchange membrane, such as Nafion, and an imposed concentration gradient were used to form an exclusion zone in a microfluidic channel. On the contrary, the exclusion zone in our system is generated using a dead-pore geometry and the interfacial chemistry of the microfluidic device. Most importantly, we do not use any salt in our system to generate the exclusion zone and it is developed by the interaction of the PDMS wall in the pores with DI water. The observation of such an exclusion zone in systems without the addition of any background electrolyte has significant implications for experiments done in PDMS chips. We show that the surface chemistry of the channel walls should be considered when investigating diffusiophoretic effects as this can induce significant exclusion zones.

Through experimental observations and theoretical predictions, we verified the dependence of the thickness of the exclusion zone in the main channel on the Sh number. The difference between the theoretical predictions for the exclusion zone and the experimental observations is explained by a significant difference in ionic and particle diffusion coefficients. The maximum thickness of the exclusion zone in our setup was in the range of 150  $\mu$ m along the channel length which is comparable with previous studies[110]. The effect of the induced concentration gradient is observed in a relatively short time as a result of the limited amount of ions present in the pores,

meaning that the exclusion zone vanishes over time and the effect of the self-induced gradients is specifically significant at the initial stages of an experiment. Additionally, the particle size has an effect on the temporal stability of the depletion zone, indicating that the size of particles plays a key role in the transient dynamics of the exclusion zone. We find that the sustainability of the exclusion zone for 2  $\mu$ m silica particles is longer than 1  $\mu$ m silica particles due to the smaller diffusivity of 2  $\mu$ m silica particles based on the Stokes-Einstein equation.

The results of this study suggest potential applications in the micro-scale separation of colloidal particles. This self-generated exclusion zone using dead-end pore geometry provides a low-cost, energy-efficient, and easy-portable solution in membrane-less water filtration.

# **Chapter 4**

# 4 The diffusiophoretic motion of porous and non-porous silica particles in a dead-end pore microfluidic channel

In this chapter, we have studied the diffusiophoretic motion of non-porous and porous silica particles under an electrolyte concentration gradient. The aim is to, for the first time, experimentally study the diffusiophoretic behavior of porous particles. For this purpose, we compared the behavior of porous particles and non-porous particles with identical surface chemistry and in similar conditions to analyze the effective parameters controlling the phoretic motion of porous particles under an electrolyte concentration gradient. In contraction to the previous chapter, the influence of the PDMS interaction with the solution is negligible in the case of electrolyte concentration gradient as the concentration of the available ions in the aqueous solution is significantly higher than the ions created by dissolved  $CO_2$  in water. While for the non-electrolyte concentration gradient, the effect of PDMS interaction with the aqueous solution can be important because in non-electrolyte the only effective phenomenon is chemiphoresis which is small compared to the electrophoretic contribution. The PDMS interaction with the aqueous solution can indeed create an electrophoretic contribution as discussed in chapter 3.

#### 4.1 Introduction

Diffusiophoresis refers to the spontaneous motion of suspended particles driven by a solute concentration gradient. This phenomenon was first described by Derjaguin et al. for nonelectrolytes in the 1940s; then, the analysis was further extended for electrolytes by Anderson and Prieve in the 1980s [44]–[46]. The physical origin of this phenomenon for electrolytes can be broken down into two components: chemiphoresis and electrophoresis. Typically, the presence of a chemical gradient produces local osmotic pressure variations along a particle surface causing particles to move. Simultaneously, a gradient of anions and cations with different diffusivities develops a local electric field to maintain electroneutrality in the solution that drives charged particles into motion. Similar physics applies for non-electrolyte solely through the action of chemiphoresis [29], [57], [58], [127]. Diffusiophoresis has been utilized in many various

applications in recent years such as drug delivery, wastewater treatment, separation, enhanced oil recovery, and laundry detergency [50], [52], [54], [55], [110].

During the past decades, the diffusiophoretic motion of charged non-porous particles has been investigated in a wide range of theoretical, numerical, and experimental aspects [131]–[134], [84], [68], [30], [29]. The diffusiophoretic velocity of a non-porous particle in a Newtonian fluid is defined as[58]

$$\boldsymbol{U}_{DP} = D_{DP} \boldsymbol{\nabla} ln \mathcal{C} \quad (4.1)$$

Here,  $C(\mathbf{x},t)$  is the solute concentration as a function of position and time and  $D_{DP}$  is the diffusiophoretic mobility of particles which in the case of a thin Debye layer (Debye length  $\ll$  radius of the particle) and symmetric electrolytes, is given by[67]

$$D_{DP} = \frac{\varepsilon k_B T}{\mu z e} \left[ \beta \zeta_p + \frac{4k_B T}{z e} ln \left( \cosh\left(\frac{z e \zeta_p}{4k_B T}\right) \right) \right]$$
(4.2)

In this equation,  $\varepsilon$  is the electrical permittivity of the solution,  $k_B$  is the Boltzmann constant, T is the absolute temperature,  $\mu$  is the viscosity of the solution, z is the valence of the ions, e is the elementary charge,  $\zeta_p$  is the zeta potential of particles, and  $\beta = \frac{D_+ - D_-}{D_+ + D_-}$  is the relative ratio of the diffusivities of cations  $(D_+)$  and anions  $(D_-)$  in the solution. The first term in eq (4.2) corresponds to the electrophoresis origin and the second term originates from chemiphoresis [67].

Diffusiophoresis of other charged colloidal entities such as porous particles has been explored in limited studies theoretically and numerically [39], [43], [40], [41]. Charged porous particles such as proteins, DNAs, porous silica, etc are important materials in various applications such as drug delivery, wastewater treatment, and separation[135], [136]. Also, diffusiophoresis occurs very often in these applications where charged non-porous or porous particles are present in these systems. In general, the migration of porous particles is faster than non-porous particles as a result of two effects. First, the porosity of porous particles with fully interconnected pores reduces the overall hydrodynamic drag force felt by porous particles because the suspending medium can flow through the particle to some degree which causes in increasing the particles' velocity in the solution[38], [41], [43]. Second, the fixed charges on the surface of porous particles are distributed all over the entire interior pore area of the particle creating a large electric driving force to attract

more counterions into the particle interior which can influence the nominal charge of porous particles[38], [42]. In the past decade, various numerical studies have shown that with increasing the porosity of porous particles, the diffusiophoretic velocity of particles increases as well. Also, the effect of double-layer thickness, fixed charges on the surface of porous particles, and the relative ratio of the diffusivities of ions ( $\beta$ ) on the diffusiophoretic mobility has been investigated theoretically [38], [41], [43]. The diffusiophoretic velocity for porous particles is obtained as[43]

$$\boldsymbol{U}_{DP} = \frac{\varepsilon}{\mu} \frac{|\nabla C|}{C_0} \left(\frac{kT}{Ze}\right)^2 \left[ (\kappa a)^2 \beta A_1 \bar{Q} + (\kappa a)^4 A_2 \bar{Q}^2 \right] + O(\bar{Q}^3)$$
(4.3)

Where  $\varepsilon$  is the dielectric permittivity,  $C_0$  is prescribed electrolyte concentration,  $\kappa^{-1}$  is the Debye length, a is the radius of the colloidal particle,  $\bar{Q} = \frac{ZeQ}{\varepsilon\kappa^2 kT}$  is a dimensionless fixed-charge density,  $A_1$  and  $A_2$  are functions of permeability, the volume fraction of particles, and the electrokinetic radius ( $\kappa a$ ). This is a solution to the linearized Poisson-Boltzmann equation where higher order terms are neglected ~  $O(\bar{Q}^3)$ .

To the best of our knowledge, there is currently no experimental study on the phoretic motion of porous particles under a concentration gradient. In this work, we experimentally investigate the difference in the migration of charged porous silica particles and non-porous silica particles under a temporary concentration gradient in dead-end pore geometries as the first experimental study on this topic. We analyze the phoretic motion of porous particles under an imposed electrolyte concentration gradient to distinguish their diffusiophoretic behavior from non-porous particles. This will help in designing an efficient experimental setup to benefit from the useful properties of porous particles, the larger electric driving force and reduced hydrodynamic drag force, for different applications like sorting or delivering particles in micro-scale systems.

#### 4.2 Experimental setup

#### 4.2.1 PDMS microfluidic channel

We use a microfluidic chip with a configuration shown in Fig 4.1. We used a dead-end pore geometry which consists of two sections, the main channel, and pores. The width and the height of the main channel are 300  $\mu$ m and 90  $\mu$ m, respectively. The height, length, and width of pores are 30  $\mu$ m, 1000  $\mu$ m, and 50  $\mu$ m, respectively. The narrow pore geometry (width << length) allows

for neglecting the pressure gradient in the y-direction[30]. Additionally, the height of dead-end pores is thinner than the main channel to reduce the disturbance from the main channel inside pores[85]. Therefore, this channel design can be considered as a one-dimensional motion of colloidal particles in the x-direction under the effect of the transient concentration gradient.



Fig 4.1 Schematic illustration of the dead-end pore geometry and dimensions. The pores are filled with solution 1 ( electrolyte solution), then, solution 2 which is particle suspension in DI water is entered into the channel.

The PDMS channels are prepared using soft lithography procedures by mixing polydimethylsiloxane (PDMS) with a curing agent (Dow Corning, Sylgard<sup>®</sup> 184) in a ratio of 10:1 by weight. Subsequently, the mixture goes through a desiccation process in the vacuum desiccator to remove air bubbles inside the mixture. Then, we pour the mixture on a patterned silicon wafer mold and put the master mold in the vacuum desiccator to de-bubble it. Afterward, we put the master mold in the heater for 2 hours at 90 °C. In the final step, we peel back the cast PDMS and plasma-bond it to a glass slide.

#### 4.2.2 Salt solutions and particle suspension

We used NaCl (Sigma-Aldrich) and NaOH (Sigma-Aldrich) salts in the experiments to create the concentration gradient of electrolytes. Initially, salts were dissolved in DI water to make 100 mM stock solutions; the solutions were further diluted to 10 mM for doing experiments.

We used 1  $\mu$ m non-porous silica particles and 1  $\mu$ m porous silica particles (SOLAD<sup>TM</sup> particle, Glantreo) with two different pore sizes, 2 nm (BET surface area = 835.69 m<sup>2</sup>/g), and 10 nm (BET surface area = 300.69 m<sup>2</sup>/g). The silica particles are suspended in DI water using a bath sonicator for ~20 min at a concentration of 2.6 × 10<sup>-4</sup> by volume (dilute regime of concentration).

The zeta potential of particle suspensions is measured using a Zetasizer nano (Malvern). The measurements are carried out using the commercial software provided by Malvern.

#### **4.2.3** Experimental protocol for microfluidic channel

All the channels are used after 24 h of plasma-bonding it to a glass slide to minimize the effects of changes in PDMS surface properties[137]. In each experiment in the microfluidic channel, we initially inject the 10 mM salt solution by using a syringe pump (Harvard Apparatus) into the channel. Once it has filled the main channel, we block the outlet and apply gentle pressure to gradually fill the pores. After filling all pores with salt solution, the plug blocking the outlet is removed and the main channel is evacuated by introducing an air plug. Then, the particle suspension is flown through the main channel with a flow rate of 250  $\mu$ L/h. As the particle suspension reaches the entrance of the first pore, the rate of injection is reduced to 30  $\mu$ L/h. The particles are visualized with an inverted microscope (ECHO, RVL-100-G, San Diego CA, USA) at 20× magnification and their motion is recorded under the microscope for almost 600 s. General image analysis is carried out in ImageJ.

#### 4.2.4 Experimental protocol for drag force measurement

To study the hydrodynamic drag force difference between non-porous and porous particles, we measured the drag force of 2  $\mu$ m non-porous silica particles and 2  $\mu$ m porous silica particles (SOLAD<sup>TM</sup> particle, Glantreo) with a pore size of 2 nm (BET surface area = 1221.7322 m<sup>2</sup>/g) and 100 nm (BET surface area = 24.711 m<sup>2</sup>/g) using an optical tweezers instrument (Tweez 305, Aresis). Not that these are different particles than the particles used for the dead-end pore experiments. The silica particles are suspended in DI water using a bath sonicator at a very dilute regime (less than 10<sup>-5</sup> volume fraction). Then, we put a small volume of particle suspension on a

block area created by paraffin on a microscope slide and cover it with a coverslip. The measurements are done with a trap with a total strength of 0.1. Afterward, the drag force is measured by exerting different constant velocities on the particle in the range of 0 to  $20 \,\mu ms^{-1}$ .

#### 4.3 **Results and discussion**

#### 4.3.1 Drag force measurement using optical tweezers

As mentioned earlier, the hydrodynamic drag force is reduced for porous particles due to their permeability of the fluid through the interior pore area[38]. In our setup, the measured drag force of charged porous and non-porous silica particles ( $\zeta_p = -37 \text{ mV}$ ) using an optical tweezers instrument is shown in Fig 4.2. We investigate the effect of surface area on drag force by measuring the drag force for porous silica particles with two different pore sizes (P1=2 nm, P2=100 nm) and non-porous silica particles. Stokes' drag force equation for non-porous particles verifies the obtained drag force for non-porous silica particles as follows

$$\mathbf{F} = 6\pi\mu a \mathbf{V} \qquad (4.4)$$

Where  $\mu$  is the fluid viscosity which is the viscosity of DI water at 22 °C (0.00095 kg.m.s<sup>-1</sup>) in this study as the particles' concentration is in the dilute regime, a is the particle radius, and **V** is the exerted velocity on the silica particle. In Fig 4.2, the experimental slope for non-porous silica particles is 0.0192 mg/s which shows a good agreement with the theoretical slope using eq (4.4) (0.018 mg/s) having a relative error of less than 7 percent.



Fig 4.2 Drag force of 2  $\mu$ m non-porous silica particles and porous silica particles as a function of velocity using the optical tweezers. Linear fits to represent Stokes' law are included (dashed lines). Also, the effect of the ratio of pore size to water molecule size is presented. The green line representes the non-porous drag force where the ratio of pore size to water molecule size is zero. The blue line represents 2  $\mu$ m porous particles with 2 nm pores where the ratio of pore size to water molecule size is 7.4. The yellow line shows 2  $\mu$ m porous particles with 100 nm pores where the ratio of pore size to water molecule size is 370.

As can be seen in Fig 4.2, the drag force decreases significantly with increasing the pore size from 2 nm to 100 nm. The drag force of particles with larger pores is 50% less than the small-pore particles and approximately 65% less than non-porous particles. Although the surface area of the particles with small pores is significantly larger than that of the large pore particle, the overall drag force of the small pores is larger compared to the particles with bigger pores, which is in line with expectations. We postulate this behavior arises from the ratio of pore size to water molecule size because it can affect the permeability of the fluid through the interior region of porous particles. Therefore, the pore size is more effective than the available surface area in changing the permeability in our experiments.

As the ratio of pore size to water molecule size increases from 7.4 to 370, the drag force reduces by 50 percent as shown in Fig 4.2. This represents the ratio of the particle's pore size to the solute molecule is a key parameter affecting the permeability and the overall hydrodynamic drag force. This ratio for the particles with 2 nm pores is 7.4 which is closer to the non-porous particles than the particles with 100 nm pores where the ratio is 370. The difference in the drag force of the particles with 2 nm pores with non-porous particles' drag force is almost 18%.

#### **4.3.2** Diffusiophoretic motion of non-porous particles

To study the effect of the interior structure of colloidal particles on the diffusiophoretic motion of particles, we use a dead-end pore geometry to create a temporary concentration gradient in the pores. As shown in eq (4.2),  $\beta$  and particles' zeta potential are the effective parameters to change the diffusiophoretic mobility of non-porous particles. Both  $\beta$  and  $\zeta_p$  can be positive or negative based on the available ions in the solution and the particle surface charge. Considering the attractive interaction between the particle and the solute molecule in our system, when  $\beta \zeta_p > 0$ , non-porous colloidal particles migrate toward higher electrolyte concentration. Otherwise, non-porous particles tend to move along lower concentrations of electrolyte. In our work,  $\beta$  for sodium chloride and sodium hydroxide solution is negative meaning that negatively charged non-porous particles are expected to move toward the higher solute concentration as shown in Fig 4.3.



Fig 4.3 Schematic illustration of negatively charged particles under the electrolyte concentration gradient. In this figure, the pore is filled with the electrolyte which the diffusivity of negative ions is higher than positive ions, and the particle suspension without any electrolyte is passing over the entrance of the pore. At the starting point (t = 0), the particles just reach the entrance of the pores. After reaching the entrance of the pore, they will migrate toward the inside of the pore under the influence of electrolyte concentration gradient. The mechanism happening at the entrance of the pore around the negatively charged particle is shown in the right hand side for the NaCl electrolyte as an example, where the slower effective motion of Na<sup>+</sup> results in an electric field that drives the negatively charged particles into the pore.

Moreover, a salt with higher  $\beta$  induces faster diffusiophoretic motion of non-porous particles, according to eq (4.2). As  $\beta$  increases, both the electrophoretic term and the chemiphoretic term increase resulting in enlarging the diffusiophoretic mobility. In Fig 4.4 (a) & (b), the migration of charged non-porous silica particles under the concentration gradient of NaCl (10mM,  $\beta$  = -0.21) and NaOH (10mM,  $\beta$  = -0.6) is shown, respectively.

As we expected, the non-porous silica particles in both electrolytes are moving into the pores over time due to the diffusiophoretic velocity as illustrated in Fig 4.3. The net observed particle motion in the pores is a combination of diffusiophoresis and wall-driven diffusioosmosis. The diffusioosmotic flow of the aqueous phase is created along the pore walls because the channel walls (PDMS) are charged[30], [55], [67] which is in the negative x-direction to maintain the zero net mass flux condition in the pores. As the zeta potential of the PDMS wall increases, the diffusioosmotic velocity increases based on recent studies[138], [139]. Furthermore, the distribution of 1  $\mu$ m non-porous silica particles at the maximum position in the pores is approximately parabolic in the 10 mM NaOH as is clearly shown in Fig 4.4(b); however, in Fig 4.4(a), the front of particles is nearly flat. The shape of the distribution at the front shows the effect of the diffusioosmotic velocity on the particles' migration. The maximum position is defined as the maximum location of migrated particles inside the pores where there are more than 1 percent of total particles are present in that location.



Fig 4.4 Entrainment of 1  $\mu$ m non-porous silica particles into the pores in the presence of the electrolyte concentration gradient as a function of time. (a) Migrating of 1  $\mu$ m non-porous silica particles under the NaCl concentration gradient. (b) The motion of 1  $\mu$ m non-porous silica particles under the NaOH concentration gradient.

It can be observed that the diffusioosmotic velocity in the 10 mM NaOH (pH~12) solution is larger than the 10 mM NaCl (pH~7) solution which is in line with expectation as the magnitude of wall zeta potential in 10 mM NaOH is larger than the 10 mM NaCl solution.

The dimensionless maximum position of non-porous silica particles over time under the concentration gradient of NaCl and NaOH is displayed in Fig 4.5. The maximum position of particles is scaled by the entrance length of the pores. The entrance length can be defined as the length from the inlet of the pores where the flow is hydrodynamically developing inside the pores. An empirical equation for calculating the entrance length in a rectangular cross-section is[94]

$$L_e = w_{cs} \left( \frac{0.55}{1 + 0.13Re} + 0.065Re \right)$$
(4.5)

Where  $L_e$  is the entrance length,  $w_{cs}$  is the width of the cross-section and Re is the Reynolds number which is defined as Re =  $\rho u w_{cs} / \mu$  for a rectangular channel in which  $w_{cs}$  is the larger side. In our setup, the entrance length is 27.5  $\mu$ m using eq (4.5).



Fig 4.5 Dimensionless maximum position of 1  $\mu$ m non-porous silica particles under the imposed electrolyte concentration gradient over time. The red squares and blue circles represent the influence of 10 mM NaOH and 10 mM NaCl, respectively.

As shown in Fig 4.5, the scaled maximum position of 1  $\mu$ m non-porous silica particles under the concentration gradient of 10 mM NaOH is larger than the 10 mM NaCl case. This can be explained through the relative diffusivities of the ions,  $\beta$ , in the system according to eq (4.2). As  $\beta$  increases, the electrophoretic contribution increases as well resulting in higher diffusiophoretic mobility. The  $\beta$  value for NaCl and NaOH is -0.21 and -0.6, respectively, and thus it is expected that the particles move further in a NaOH concentration gradient when compared to the same NaCl concentration gradient.

Moreover, the presence of electrolytes influences the zeta potential of particles which is the second effective parameter in the diffusiophoretic mobility as shown in eq (4.2). The changes in the particle's zeta potential affect both the electrophoretic and chemiphoretic terms in eq (4.2). Based on the zeta potential measurements, the zeta potential of 1 $\mu$ m non-porous silica particles in the 10 mM NaCl solution and 10 mM NaOH solution is -20.57 ± 1.91 mV and -98.43 ± 4.89 mV, respectively.

We calculate the diffusiophoretic mobility of 1µm non-porous silica particles based on eq (4.2) using values of the physical parameters in our system which results in  $D_{DP, NaCl} = 2.4 \times 10^{-10} \text{ m}^2\text{s}^{-1}$  and  $D_{DP, NaOH} = 12.3 \times 10^{-10} \text{ m}^2\text{s}^{-1}$ . Therefore, the diffusiophoretic velocity in the 10 mM NaOH is approximately 5 times larger than 10 mM NaCl which leads non-porous particles to move faster and reach a larger maximum position compared to the 10 mM NaCl solution which is in agreement with experimental results shown in Fig 4.5. Also, these results are in line with Shim et al. study about the effect of zeta potential and electrolyte on the diffusiophoretic mobility[139]. The migration of non-porous particles into the pores is getting slower over time due to the reduction of the electrolyte concentration gradient inside the pores which decreases the diffusiophoretic motion.

#### **4.3.3** Diffusiophoretic motion of porous particles

We visualize the diffusiophoretic motion of 1  $\mu$ m porous silica particles with two different pore sizes (2 nm and 10 nm) under the same electrolyte concentration gradients as the non-porous particles. The observation for both samples of porous particles under the concentration gradient of 10 mM NaCl is shown in Fig 4.6. In Fig 4.6(a), the migration of 1  $\mu$ m porous silica particles with 2 nm pore size shows that a very limited number of particles enter the pore compared to the same

condition for 1 µm non-porous silica particles (Fig 4.4 (a)). Also, for the particles with 10 nm pores, particles just enter close to the entrance of the pores over time as displayed in Fig 4.6(b), but they do not move further toward the inside of the pores. This expected behavior arises due to the two main effects. First, when the porous particle suspensions are in contact with the 10 mM NaCl at the entrance of the pores, the Debye layer is very thin ( $\kappa^{-1} \sim 3$  nm) which means  $\kappa a$  is large (~ 160). When  $\kappa a$  is large, the fixed charges carried by the porous particles can be neutralized by driving counterions into the interior region of porous particles. This phenomenon is called "counterion condensation" or "shielding effect" which causes porous particles to be chargeless and suppress the chemiphoretic contribution completely[38], [140]–[142]. Therefore, the second term in eq (4.3) will be zero. It should be noted that this behavior is in direct contrast to the behavior of non-porous particles for which the chemiphoretic contribution increases with increasing  $\kappa a$ .



Fig 4.6 Diffusiophoretic motion of porous silica particles under the concentration gradient of 10 mM NaCl. (a)Time sequence images for 1  $\mu$ m porous silica particles with 2nm pores (b) Time sequence images for 1  $\mu$ m porous silica particles with 2nm pores (b) Time sequence images for 1  $\mu$ m porous silica particles with 2nm pores.

Additionally, the diffusiophoretic mobility for porous particles as shown in eq (4.3) is a function of the dimensionless particle surface charge density. The measured zeta potential for 2 nm and 10

nm pore size particles in 10 mM NaCl is  $-22.66 \pm 1.95$  mV and  $-23.43 \pm 0.9$  mV, respectively. The surface charge density for a binary system can be determined using the Gouy-Chapman equation which relates the particle zeta potential to the surface charge density as shown below[143].

$$Q = \frac{2\varepsilon k_B T}{\kappa^{-1} e} \sinh\left(\frac{e\zeta_p}{2k_B T}\right) \quad (4.6)$$

According to eq (4.6) and the dimensionless surface charge equation, the dimensionless surface charge for 2 nm and 10 nm pore size particles in 10 mM NaCl is  $-7.815 \times 10^{-5}$  and  $-8.1 \times 10^{-5}$ , respectively. The actual magnitude of the surface charge is higher than this amount because the zeta potential is measured at the slipping plane which has a lower electric potential than the surface. Based on recent studies, these values for surface charge density are very low[38]–[41], [43] resulting in near-zero diffusiophoretic mobility.

Second, the imposed velocities on the particles at the entrance of the pores in the x-direction and y-direction are diffusiophoretic velocity and convective velocity, respectively as represented in Fig 4.3. The reduced hydrodynamic drag force of the porous particles in the main channel causes the effect of convective velocity in the y-direction to increase.

As a result, the small value in the surface charge density creates a very small diffusiophoretic velocity, and the reduced hydrodynamic drag force in the main channel increases the particles' convective velocity in the main channel. The balance between these two velocities will be dominated by the convective velocity and porous particles tend to move in the main channel without being captured into the pores. For capturing porous particles inside the pores, the porous particles' dimensionless surface charge density should be larger than 0.01[41], [43].



Fig 4.7 Scaled maximum position of 1  $\mu$ m non-porous and porous silica particles under the 10 mM NaCl concentration gradient over time. The orange circles, green triangles, and red squares represent particles with pores of 2 nm, 10 nm, and non-porous particles respectively.

The scaled maximum position over time for porous particles with small 2 nm and 10 nm pores under the concentration gradient of NaCl is shown in Fig 4.7. In this figure, the scaled maximum position of particles with 10 nm pores does not exceed 1 over time. This means porous particles did not diffuse more than the entrance length into the pore and they are solely migrating by the hydrodynamic flow which creates vortexes at the inlet of the pores. Also, a limited number of particles with 2 nm pores in 10 mM NaCl are moving inside the pores over time as shown in Fig 4.6(a). This behavior stems from the small pore size of porous particles which leads to behaving closer to non-porous particles in the solution which results in the migrating of some of the 2 nm pore size sample inside the pore. Therefore, a limited number of them are migrating inside the pores under the 10 mM NaCl concentration gradient. The scaled maximum positions displayed in Fig 4.7 for 2 nm pore particles are almost the same as non-porous particles under the effect of 10 mM NaCl.

The diffusiophoretic motion of charged porous silica particles under the concentration gradient of NaOH (10 mM) is shown in Fig 4.8. As NaOH is a strong base, it changes the pH inside the pores and a pH gradient is created along the pore in addition to the electrolyte gradient. The pH gradient can change the surface charge of silica particles significantly. As shown in Fig 4.8, both samples of porous particles do not enter inside the pores of the channel, and they are creating a small depletion zone over the entrance of the pores. This means a diffusiophoretic velocity in the negative x-direction is imposed on them which is in the opposite direction of the case of the non-porous particle.



Fig 4.8 Diffusiophoretic motion of porous silica particles under the concentration gradient of 10mM NaOH. (a) Time sequence images for 1  $\mu$ m porous silica particles with 2 nm pore size. (b) Time sequence images for 1  $\mu$ m porous silica particles with 10 nm pore size.

All conditions for non-porous and porous particles in the case of 10 mM NaOH are the same; however, non-porous particles migrate inward the channel pores and porous particles migrate outward the channel pores, forming a small exclusion zone of particles over the pores. The only parameter that can be different is the sign of the effective surface charge of porous particles compared to non-porous particles in the presence of NaOH electrolytes. Also, a similar behavior for porous particles has been observed in a 5 mM NaOH solution.

The zeta potential measurements of porous particles in DI water and 10 mM NaOH solution are shown in Table 4.1. The porous particles' zeta potential in 10 mM NaOH is similar to the zeta potential of porous particles in DI water, while for the non-porous particles, a significant difference in zeta potential is observed in the electrolyte solution.

Table 4.1 Zeta potential measurements of non-porous and porous particles with DLS instrument in DI water and 10 mM NaOH solution.

Particle sample	Zeta potential in	Zeta potential in
	DI water	10 mM NaOH
Porous silica with 2nm pores	$-35.62 \pm 0.94 \text{ mV}$	$-41.72 \pm 3.53 \text{ mV}$
Porous silica with 10nm pores	$-42.2\pm0.45~mV$	$-42.7 \pm 1.77 \text{ mV}$
Non-porous silica	$-51.73 \pm 0.71 \text{ mV}$	$-98.43\pm4.89\ mV$

The zeta potential is the electric potential at the location of the slipping plane in the electric double layer but the diffusiophoretic mobility for porous particles is a function of the surface charge of porous particles, which can be significantly different from each other. In the high pH created by 10 mM NaOH, we hypothesize counterions (positive ions) are attracted to the surface of porous particles (the interior region and outer region) and make a temporary positive surface charge for porous particles. Therefore, porous particles' effective surface charge is positive temporarily resulting in migration in the negative x-direction and creating a temporary small depletion zone near the entrance of the pores[139]. After the concentration gradient of NaOH disappears (~300s), the only imposed velocity on porous particles is the convective velocity in the y-direction and the surface charge of particles will become negative again. Therefore, the generated small depletion zone disappears by the Brownian motion of particles in the y-direction as mentioned in chapter 3.

#### 4.4 Conclusions

Porous particles can be used in various applications such as separation and drug delivery where a concentration gradient of ions can be present or be developed as desired in such systems. Among several factors that can influence the mobility of porous particles, we highlighted the influence of

pore size on the permeability, and the electrolyte concentration gradient in the range of a thin double layer.

We investigated the diffusiophoretic motion of non-porous and porous silica particles in a deadend pore channel in two different electrolytes experimentally. The porous silica particles show a lower drag force up to 65% compared to the non-porous silica particles in the solution using an optical tweezers instrument to measure the drag force. The pore size of porous particles has a significant influence on the overall drag force of particles. Although the particles with 2 nm pores in our experiments have a larger surface area compared to particles with a pore size of 100 nm, the measured drag force for the particles with larger pores was 50% lower than that of particles with smaller pores.

Also, we observe that sodium hydroxide salt leads to faster diffusiophoretic motion in comparison with sodium chloride for non-porous silica particles due to the higher zeta potential of particles in NaOH and higher  $\beta$  than NaCl. These two parameters are playing a main role in the diffusiophoretic mobility of non-porous particles as shown in eq (4.2). On the other hand, the diffusiophoretic mobility of porous particles is more complex than non-porous particles. The surface charge density,  $\beta$ ,  $\kappa a$ , and surface area affect the diffusiophoretic mobility of porous particles. Under the concentration gradient of NaCl and NaOH, the chemiphoretic contribution reduces to zero due to the high value of  $\kappa a$ . Therefore, the electrophoretic contribution is the only source for creating diffusiophoretic mobility and the particle motion is dictated by the net local electric field and the effective zeta potential of the porous particles.

In the 10 mM NaCl, the small value of surface charge density and the reduced hydrodynamic drag force play crucial roles in the diffusiophoretic motion of porous particles in dead-end pore geometries. Both of these parameters prevent porous particles to move inside the pore along the concentration gradient and significantly alter the behavior of particles depending on the pore size. For particles with large pores, this effectively diminishes any diffusiophoretic movement into the dead-end channels, while particles with smaller pores are still driven into the channels. This implies that when working with porous particles in electrolyte solutions, the interaction between the electrolyte and the particle pores will dictate effective transport. When changing the salt to 10 mM NaOH, the effective surface charge of porous particles temporarily changes to a positive surface charge causing the push away of the porous particles from the entrance of the pores,

forming an exclusion zone similar to the observations in Chapter 3. This behavior under the effect of 10 mM NaOH can be utilized in the micro-scale manipulation of particles.

Our work shows that the interaction of the porous particles, the pore size, and the background electrolyte have a large influence on the diffusiophoretic behavior of porous particles, indicating that for applications the system should be studied to sufficient detail before implementation.

# **Chapter 5**

## **5** Conclusions and Recommendations

In this dissertation, the phoretic motion of non-porous and porous colloidal particles under the effect of a solute concentration gradient was investigated experimentally in a dead-end pore microfluidic channel. First, the influence of PDMS interaction with available ions in DI water on the particles' motion was studied without introducing any background concentration gradient. In this part, the effect of particle size, the injection flow rate, and different sizes for the width of the pores was explored over time. Next, the phoretic motion of non-porous and porous particles under an imposed electrolyte concentration gradient was studied. The previous studies explained the porous particles' motion under the effect of concentration gradient numerically. In this work, the diffusiophoretic behavior of porous particles was compared to non-porous particles experimentally based on the drag force measurements, the influence of surface charge density, and the porosity of porous particles as a function of pore size.

The first novel aspect of this work is to study the influence of the PDMS wall interfacial interaction with present ions in DI water without introducing any background concentration gradient in the channel. We showed the interfacial chemistry of the microfluidic platform can have a significant influence on the particles' movement and the impact of this interfacial chemistry needs to be considered in lab-on-a-chip designs. A second novel contribution is the investigation of the diffusiophoretic motion of porous particles experimentally and the comparison with the case of non-porous particles. Based on our knowledge, the migration of porous particles under the effect of an imposed concentration gradient has not been investigated experimentally in previous studies. The interior microstructure of colloidal particles can significantly change the particles' diffusiophoretic behavior. Therefore, it is crucial to study the diffusiophoretic motion of porous particles and distinguish it from the case of non-porous particles.

#### 5.1 Conclusions

First, the effect of PDMS interaction with available ions present in DI water inside the channels was found can create a transient diffusiophoretic motion in colloidal particles (chapter 3). We have

shown that in dead-end pore geometry without introducing any electrolyte into the channel a temporary spontaneous concentration gradient between the pores and the main channel would be created due to the interaction of the PDMS wall inside the pores with ions in DI water ( $H^+$ ,HCO<sub>3</sub><sup>-</sup>). This indicates that even without using ion exchange membranes and a background concentration of electrolyte, as conventionally done[31], [51], [119], we can create an exclusion zone in the channel. This self-generated concentration gradient results in the accumulation of colloidal particles in the main channel and pushes them away from the pores of the channel to create a particle-free zone in the main channel which is called an exclusion zone. The exclusion zone development in our system without introducing any background electrolyte has a significant role in the PDMS dead-end pore microfluidic channel. This effect needs to be considered at the initial stage of the experiments because it can affect the migration of colloidal particles in the channel.

We found the thickness of the generated exclusion zone depends on the Sherwood number which is the ratio of the longitudinal convective transfer to the transverse diffusion rate. As Sh increases, the thickness of the exclusion zone decreases because the convective velocity increases, and the balance of velocities on particles at the entrance of the pores will be dominated by the convective velocity. Also, it was found that the width of the pores does not affect the exclusion zone establishment as the density of available ions remains constant as the pore width changes. Furthermore, we have investigated the effect of particle size on the exclusion zone generation. The results for two different sizes of colloidal particles showed that the size does not change the thickness of the exclusion zone; whereas, it will affect the temporal stability of the exclusion zone. The diffusion coefficient of colloidal particles depends on the particle size based on the Stokes-Einstein equation. The influence of the diffusiophoretic velocity on the colloidal particles reduces over time as the concentration gradient is decreasing. Therefore, the migration of the particles in the transverse direction is getting increasingly dominated by the Brownian motion over time. This means the exclusion zone disappears in a shorter time for the smaller particles compared to larger particles because their diffusion coefficient is larger.

In chapter 4, the diffusiophoretic motion of porous particles under the influence of electrolyte concentration gradient was explored and compared with non-porous particles. We investigated the effect of pore size, electrolyte concentration gradient in the range of a thin double layer, and the relative difference in diffusivity of ions ( $\beta$ ). It should be mentioned that the effect of the interaction

of PDMS with DI water on the particles' motion can be neglected when there is a concentration gradient of electrolyte in the system because the influence of the concentration gradient of the introduced electrolyte would be dominant due to the electrophoretic contribution which is significantly higher than the chemiphoresis effect[41]. In the case of the non-electrolyte concentration gradient, the effect of the PDMS interaction with the solution should be considered because the phoretic motion of particles in the non-electrolyte concentration gradient is caused by the chemiphoresis, but the interaction of PDMS with the solution is creating the electrophoretic contribution as well.

At first, we showed the measured drag force for porous particles is up to 65% smaller than nonporous particles based on the pore size. As the pore size decreases, the measured drag force for porous particles increases and gets closer to the non-porous particles' drag force. We then studied the diffusiophoretic motion of porous and non-porous silica particles in a dead-end pore microfluidic channel. For non-porous particles, we observed the diffusiophoretic motion of particles in the NaOH concentration gradient is higher than the NaCl concentration gradient due to the higher magnitude of zeta potential of silica particles in NaOH solution and higher  $\beta$ compared to NaCl.

For porous particles, various parameters play a role in the diffusiophoretic motion such as surface charge density,  $\beta$ ,  $\kappa a$ , pore size and surface area which both can affect the permeability based on the interconnectivity of pores. One of the most important differences between porous particles and non-porous particles is the effect of surface charge density. For porous particles, the surface charge density affects the diffusiophoretic mobility; whereas, the zeta potential is the parameter controlling the diffusiophoretic mobility for non-porous particles. The zeta potential is the potential around the particles on the slipping plane which could be significantly different from the potential on the surface of porous particles because counterions can be attracted into the interior region of porous particles and affect the surface charge density.

In the 10mM concentration of NaCl or NaOH, the value of ka is large which makes the porous particles effectively chargeless. Therefore, the chemiphoretic contribution is zero and the diffusiophoretic motion of particles solely stems from the electrophoretic contribution[39], [41], [43]. Under the imposed concentration gradient by the 10 mM NaCl in the dead-end pore microfluidic channel, we found that the diffusiophoretic velocity compared to the convective

velocity is small due to the small surface charge density and the reduced hydrodynamic drag force. Therefore, porous particles cannot diffuse into the pores along the concentration gradient because both of the mentioned parameters are reducing the influence of diffusiophoretic velocity.

Under the concentration gradient of 10 mM NaOH, we observed the particles are being pushed away from the entrance of the pores over time and creating an exclusion zone over the pore entrance. We hypothesize the effective surface charge of porous particles changes to a positive surface charge temporarily because this could be the only reason causing the direction of particles' motion changes to the opposite direction. After the effect of 10 mM NaOH concentration disappears, the porous particles' surface charge returns to negative and particles will move back toward the entrance of the pores without being pushed away.

The observations in our experimental setup for non-porous and porous particles displayed that the diffusiophoretic motion of porous particles could be very complex due to the interactions of porous particles with the ions in the solution which can affect significantly the surface charge density of particles. Also, the pore size and the surface area of particles can affect the permeability of porous particles which is an effective factor in diffusiophoretic mobility. Therefore, the phoretic motion of porous particles with different properties should be studied in sufficient detail before implementation.

### 5.2 **Recommendations and Future Directions**

At the end of this research, based on the obtained results and insights from the experimental study on the diffusiophoretic motion of non-porous and porous particles in a PDMS microfluidic channel, recommendations for further research of this work can be made;

 As explained in the previous section, our work shows the behavior of porous silica particles under the effect of an electrolyte concentration gradient. However, the behavior of porous particles is very complex due to the effect of various parameters such as porosity, and porous structure which can affect the permeability, and the pore sizes. Therefore, each of these parameters needs to be explored separately to have a comprehensive understanding of the influence of each of these parameters on the diffusiophoretic motion of porous particles.

- 2. The dead-end pore geometry provides a transient concentration gradient which means the diffusiophoretic behavior is changing over time. To simplify the situation in the experimental setup for the diffusiophoretic motion of porous particles, open pores or H-shaped microfluidic channels can be used to be able to generate a steady-state concentration gradient in the channel. In these configurations, the diffusiophoretic velocity of colloidal particles can be calculated as it is constant during each concentration gradient considering the influence of diffusioosmotic velocity. Therefore, the diffusiophoretic mobility of colloidal particles can be calculated for porous particles in different conditions as a function of porosity, pore size,  $\beta$ , and surface charge density. Then, the calculated diffusiophoretic mobility based on experimental data can be compared to the numerical data shown in the previous studies.
- 3. One of the common porous particles used in drug delivery applications is polyelectrolytes such as DNA and proteins. The diffusiophoretic motion of polyelectrolytes requires to be explored and the differences between porous silica particles and polyelectrolytes should be distinguished as polyelectrolytes are macromolecules and have different porous structures compared to porous silica particles.
- 4. The presence of multiple ions in the solution is a common condition in various applications. Therefore, it is worth studying the effect of the presence of multiple ions on the diffusiophoretic behavior of colloidal particles. The diffusiophoretic motion of non-porous particles in the presence of multiple species has been explored both theoretically and experimentally[30], [80]. However, it has not been explored for porous particles theoretically or experimentally. Thus, a potential study to explore is the diffusiophoretic motion of porous particles in the presence of multiple species both numerically and experimentally.

# Appendix A

# Supplementary information for chapter 3

## A 1. Exclusion zone establishment over time.

Further pictures of exclusion zone formation in the experiment are shown in Fig. A1. As shown in Fig. A1(b), the exclusion zone thickness along the channel length increases due to the presence of pores along the channel length.



Fig. A1 Experimental observation of the exclusion zone formation in dead-end channel for pore width =  $100\mu$ m. (a) For 1  $\mu$ m silica particles at t=3min over the first pore. (b) For 2  $\mu$ m silica particles at t=3min over pores 4 to 6.
## A 2. Obtaining ions' exclusion zone thickness using boundary layer analysis.

We considered a steady-state condition for eq 2.3 in the main text to be able to obtain a solution for the exclusion zone thickness analytically as

$$D_{Heff}\frac{\partial^2 c_H}{\partial x^2} + D_{Heff}\frac{\partial^2 c_H}{\partial y^2} - \frac{\partial}{\partial y}(c_H u_y) = 0 \quad (A1)$$

Where x and y are the Cartesian coordinates and  $u_y$  is the background flow velocity in the ydirection. We use characteristic scales to nondimensionalize the above equation, the dimensionless parameters are defined as

$$c_{H}^{*} = \frac{c_{H}}{c_{H0}}, X = \frac{x}{w_{ch}}, Y = \frac{y}{L}, u_{y}^{*} = \frac{u_{y}}{\bar{u}}$$

Where  $c_{H0}$  is the bulk concentration in the pore,  $w_{ch}$  is the width of the main channel, L is the length of the main channel which includes pores and  $\bar{u}$  is the mean flow velocity of the background flow. So, eq 2.5 becomes:

$$\frac{D_{Heff}c_{H0}}{w_{ch}^2}\frac{\partial^2 c_H^*}{\partial X^2} + \frac{D_{Heff}c_{H0}}{L^2}\frac{\partial^2 c_H^*}{\partial Y^2} - \frac{c_{H0}\bar{u}}{L}\frac{\partial}{\partial Y}\left(c_H^*u_y^*\right) = 0 \quad (A2)$$

Generally, we know that  $L \gg w_{ch}$  which means the second term in eq (A2) is negligible compared to the first term. Therefore, eq (A2) can be simplified to

$$\frac{\partial^2 c_H^*}{\partial X^2} - Sh \frac{\partial}{\partial Y} \left( c_H^* u_Y^* \right) = 0 \qquad (A3) \qquad Sh = \frac{w_{ch}^2 \bar{u}}{LD_{Heff}} \qquad (A4)$$

Where Sh is the Sherwood number is the ratio of longitudinal convective transfer to the transverse diffusion rate which links most of the physicochemical parameters (flow rate, diffusivity of ions, and geometrical factors).

We propose a trial solution for the concentration profile of  $H^+$  over the first pore as a function of exclusion zone thickness to derive an analytical solution for the exclusion thickness. For this purpose, we have defined a coordinate transform of  $x = \xi - \frac{w_{ch}}{2}$  and  $y = \eta$  and establish the concentration profile in this coordinate as

$$c_{H} = \begin{cases} c_{H\infty} + c_{H0} \frac{\delta - \xi}{\delta} & \xi < \delta \\ c_{H\infty} & \xi \ge \delta \end{cases}$$
(A5)

Where  $\delta$  is the exclusion layer thickness as a function of  $\eta$  and  $c_{H\infty}$  is the bulk concentration in the main channel. Then, substituting eq (A5) in eq (A3) and integrating the differential equation could result in the von Karman integral balance as

$$D_{Heff} \left. \frac{\partial c_H}{\partial \xi} \right|_{\xi=0} + \int_0^\delta u_\eta \frac{\partial c_H}{\partial \eta} d\xi = 0 \qquad (A6)$$

As background flow is laminar and fully developed in the channel, the velocity profile over the first pore in the y-direction can be obtained through the Poiseuille flow equation between parallel plates with specific boundary conditions[108].

$$u_{y} = \frac{-\Delta p}{2\mu L} x^{2} + c_{1}x + c_{2} \qquad (A7)$$

The governing boundary conditions by considering full slip condition for the lower side and noslip condition for the upper side are:

$$\begin{cases} 1) at x = \frac{w_{ch}}{2} \rightarrow u_y = 0\\ 2) at x = -\frac{w_{ch}}{2} \rightarrow \frac{du_y}{dx} = 0 \end{cases}$$
(A8)

The final equation for flow velocity after calculating the coefficients by substituting the boundary conditions in eq (A7) would be as follows

$$u_{y} = \frac{-\Delta p}{2\mu L} x^{2} - \frac{-\Delta p w_{ch}}{2\mu L} x + \frac{3\Delta p w_{ch}^{2}}{8\mu L} = \frac{-\Delta p w_{ch}^{2}}{2\mu L} \left( \left( \frac{x}{w_{ch}} \right)^{2} + \frac{x}{w_{ch}} - \frac{3}{4} \right)$$
(A9)

Eq (A9) after substituting the coordinate transform  $(x = \xi - \frac{w_{ch}}{2})$  would be simplified to

$$u_{y} = \frac{3}{2}\bar{u}_{y}\left(\frac{5}{4} - \frac{\xi}{w_{ch}} - \left(\frac{\xi}{w_{ch}} - \frac{1}{2}\right)^{2}\right) = \frac{3}{2}\bar{u}_{y}\left(1 - \left(\frac{\xi}{w_{ch}}\right)^{2}\right)$$
(A10)

Substituting the flow velocity equation in eq (A6) and calculating the integral would be as follows if assume  $\delta \ll 2w_{ch}$ 

$$\int_{0}^{\delta} u_{\eta} \frac{\partial c_{H}}{\partial \eta} d\xi = \int_{0}^{\delta} \frac{3}{2} \bar{u}_{y} \left( 1 - \left(\frac{\xi}{w_{ch}}\right)^{2} \right) \frac{c_{H0}}{\delta^{2}} \frac{d\delta}{dy} \xi d\xi = \frac{3}{4} \bar{u}_{y} c_{H0} \frac{d\delta}{dy} \left( 1 - \frac{\delta^{2}}{2w_{ch}^{2}} \right)$$
(A11)  
$$-D_{Heff} \frac{c_{H0}}{\delta} + \frac{3}{4} \bar{u}_{y} c_{H0} \left( 1 - \frac{\delta^{2}}{2w_{ch}^{2}} \right) \frac{d\delta}{dy} = 0$$
(A12)  
$$\delta^{2} = \frac{8}{3} \frac{D_{Heff}}{\bar{u}_{y}} y$$
(A13)

## A 3. Obtaining particles' exclusion zone thickness using particles' trajectory analysis.

The particles' exclusion thickness can be obtained based on the particles' trajectory in the main channel. The fluid flow in the channel is considered to be laminar ( $\text{Re} \sim 0$ ) and the sedimentation effect is neglected. The particles' streamline on the exclusion zone boundary layer is as follows

$$\frac{dx}{dt} = \frac{d\xi}{dt} = U_{DP}$$
(A14)  
$$\frac{dy}{dt} = u_{y,max}$$
(A15)

 $u_{y,max}$  is the maximum velocity of the flow rate in the y-direction. The colloidal particles close to pores are migrating away from the pores of the channel where the flow has a no-slip boundary condition in the y-direction. The diffusiophoretic velocity equation based on the hydrogen concentration profile leads to

$$U_{DP} = D_{DP} \nabla lnC = -\frac{D_{DP}}{C_H} \frac{C_{H0}}{\delta} = -D_{DP} \frac{C_{H0}}{\delta c_{H\infty} + c_{H0}(\delta - \xi)}$$
(A16)

At the particles' streamline ( $\xi = \delta_p$ ), eq (A16) can be simplified to

$$U_{DP} = D_{DP} \nabla ln C = -\frac{D_{DP}}{C_{H\infty}} \frac{C_{H0}}{\delta}$$
(A17)

Dividing eq (A14) to eq (A15) and considering eq (A17) at the particles streamline boundary,

$$\frac{d\delta_p}{dy} = \frac{\frac{D_{DP}}{C_{H\infty}} \frac{C_{H0}}{\sqrt{\frac{8}{3} \frac{D_{Heff}}{\bar{u}_y} y}}}{\frac{3}{2} \bar{u}_y}$$
(S18)

Eq (3.11) in the main text shows that the particle exclusion thickness is a function of the inverse square root of flow velocity  $(\sqrt{\frac{1}{\overline{u}_y}})$ . As shown in Fig. A2, the ratio of normalized particles'

exclusion thickness in two different flow rates is representing the same results in the experimental data and theoretical calculation.



Fig. A2 The ratio of normalized particle exclusion thickness in two flow rates. The black curve shows the experimental value in each position. The blue and red lines display the average values over the first pore in experiment and theory, respectively.

A 4. Exclusion zone development in the main channel for different flow rates



Fig. A3 The maximum exclusion zone in the main channel for 100  $\mu$ m pore width. a) the exclusion zone development for flow rate 30  $\mu$ L/h. b) the exclusion zone development for flow rate 55  $\mu$ L/h. c) the exclusion zone development for flow rate 80  $\mu$ L/h.

As shown in Fig. A3, when the flow rate in the main channel increases, the exclusion zone thickness decreases confirming the effect of Sh described in the theoretical part.

## References

- S. Hjertén, "Free zone electrophoresis," *Chromatogr. Rev.*, vol. 9, no. 2, pp. 122–219, Dec. 1967, doi: 10.1016/0009-5907(67)80003-6.
- [2] G. S. McNab and A. Meisen, "Thermophoresis in liquids," J. Colloid Interface Sci., vol. 44, no. 2, pp. 339–346, Aug. 1973, doi: 10.1016/0021-9797(73)90225-7.
- J. R. Manning, "Diffusion in a Chemical Concentration Gradient," *Phys Rev*, vol. 124, no. 2, pp. 470–482, Oct. 1961, doi: 10.1103/PhysRev.124.470.
- [4] T. Shoji *et al.*, "Plasmonic Manipulation of DNA using a Combination of Optical and Thermophoretic Forces: Separation of Different-Sized DNA from Mixture Solution," *Sci. Rep.*, vol. 10, no. 1, p. 3349, Feb. 2020, doi: 10.1038/s41598-020-60165-5.
- [5] D. Vigolo, R. Rusconi, H. A. Stone, and R. Piazza, "Thermophoresis: microfluidics characterization and separation," *Soft Matter*, vol. 6, no. 15, p. 3489, 2010, doi: 10.1039/c002057e.
- [6] M. Jotkar and L. Cueto-Felgueroso, "Particle Separation through Diverging Nanochannels via Diffusiophoresis and Diffusioosmosis," *Phys Rev Appl.*, vol. 16, no. 6, p. 064067, Dec. 2021, doi: 10.1103/PhysRevApplied.16.064067.
- [7] L. Zhang, F. Dang, and Y. Baba, "Microchip electrophoresis-based separation of DNA," J. Pharm. Biomed. Anal., vol. 30, no. 6, pp. 1645–1654, Jan. 2003, doi: 10.1016/S0731-7085(02)00510-1.
- [8] N. Singh, G. T. Vladisavljević, F. Nadal, C. Cottin-Bizonne, C. Pirat, and G. Bolognesi, "Enhanced Accumulation of Colloidal Particles in Microgrooved Channels via Diffusiophoresis and Steady-State Electrolyte Flows," *Langmuir*, vol. 38, no. 46, pp. 14053– 14062, Nov. 2022, doi: 10.1021/acs.langmuir.2c01755.
- [9] R. Fuiko, E. Saracevic, I. J. Koenka, P. C. Hauser, and J. Krampe, "Capillary electrophoresis for continuous nitrogen quantification in wastewater treatment processes," *Talanta*, vol. 195, pp. 366–371, Apr. 2019, doi: 10.1016/j.talanta.2018.11.056.
- [10] T. Poiger, S. D. Richardson, and G. L. Baughman, "Identification of reactive dyes in spent dyebaths and wastewater by capillary electrophoresis-mass spectrometry," *J. Chromatogr. A*, vol. 886, no. 1–2, pp. 271–282, Jul. 2000, doi: 10.1016/S0021-9673(00)00482-9.

- [11] Z. Zheng, J. Zhu, S. Xia, T. Laminger, and W. Höflinger, "Separation of Small Particles by Diffusio- and Thermophoresis in a Gas-Liquid Cross-Flow Array," *Chem. Eng. Technol.*, vol. 41, no. 1, pp. 61–69, Jan. 2018, doi: 10.1002/ceat.201700133.
- [12] F. Yang, S. Shin, and H. A. Stone, "Diffusiophoresis of a charged drop," *J. Fluid Mech.*, vol. 852, pp. 37–59, 2018, doi: 10.1017/jfm.2018.531.
- [13] M. Tsai *et al.*, "Analytical solution to dielectric droplet diffusiophoresis under Debye–Hückel approximation," *ELECTROPHORESIS*, vol. 43, no. 3, pp. 495–500, Feb. 2022, doi: 10.1002/elps.202100264.
- [14] Y. Wu and E. Lee, "Diffusiophoresis of a highly charged soft particle normal to a conducting plane," *ELECTROPHORESIS*, vol. 42, no. 21–22, pp. 2383–2390, Nov. 2021, doi: 10.1002/elps.202100052.
- [15] L. Fan and E. Lee, "Diffusiophoresis of a highly charged conducting fluid droplet," *Phys. Fluids*, vol. 34, no. 6, p. 062013, Jun. 2022, doi: 10.1063/5.0098144.
- [16] X. Cheng, D. Long, L. Chen, J. A. Jansen, S. C. G. Leeuwenburgh, and F. Yang, "Electrophoretic deposition of silk fibroin coatings with pre-defined architecture to facilitate precise control over drug delivery," *Bioact. Mater.*, vol. 6, no. 11, pp. 4243–4254, Nov. 2021, doi: 10.1016/j.bioactmat.2021.03.046.
- [17] H. Qi *et al.*, "Electrophoretic Deposition of Bioadaptive Drug Delivery Coatings on Magnesium Alloy for Bone Repair," *ACS Appl. Mater. Interfaces*, vol. 11, no. 8, pp. 8625– 8634, Feb. 2019, doi: 10.1021/acsami.9b01227.
- [18] J. Xi, X. A. Si, and R. Gaide, "Electrophoretic Particle Guidance Significantly Enhances Olfactory Drug Delivery: A Feasibility Study," *PLoS ONE*, vol. 9, no. 1, p. e86593, Jan. 2014, doi: 10.1371/journal.pone.0086593.
- [19] C. M. Proctor *et al.*, "Electrophoretic drug delivery for seizure control," *Sci. Adv.*, vol. 4, no. 8, p. eaau1291, Aug. 2018, doi: 10.1126/sciadv.aau1291.
- [20] S. E. Ghasemi, "Thermophoresis and Brownian motion effects on peristaltic nanofluid flow for drug delivery applications," J. Mol. Liq., vol. 238, pp. 115–121, Jul. 2017, doi: 10.1016/j.molliq.2017.04.067.
- [21] R. Zhou, F. Shen, J. Cui, Y. Zhang, H. Yan, and S. Juan Carlos, "Electrophoretic Deposition of Graphene Oxide on Laser-Ablated Copper Mesh for Enhanced Oil/Water Separation," *Coatings*, vol. 9, no. 3, p. 157, Feb. 2019, doi: 10.3390/coatings9030157.

- [22] S. Sakthivel and M. Elsayed, "Enhanced oil recovery by spontaneous imbibition of imidazolium based ionic liquids on the carbonate reservoir," J. Mol. Liq., vol. 340, p. 117301, Oct. 2021, doi: 10.1016/j.molliq.2021.117301.
- [23] E. Lee, "Diffusiophoresis of Liquid Droplets and Gas Bubbles," in *Interface Science and Technology*, vol. 26, Elsevier, 2019, pp. 359–384. doi: 10.1016/B978-0-08-100865-2.00017-5.
- [24] Y. Pawar, Y. E. Solomentsev, and J. L. Anderson, "Polarization Effects on Diffusiophoresis in Electrolyte Gradients," *J. Colloid Interface Sci.*, vol. 155, no. 2, pp. 488–498, Feb. 1993, doi: 10.1006/jcis.1993.1063.
- [25] R. F. Stout and A. S. Khair, "Influence of ion sterics on diffusiophoresis and electrophoresis in concentrated electrolytes," *Phys Rev Fluids*, vol. 2, no. 1, p. 014201, Jan. 2017, doi: 10.1103/PhysRevFluids.2.014201.
- [26] B. K. Kalita and R. Choudhury, "The eventuality of diffusiophoresis and chemical reaction in a flow with variable fluid properties through an upright channel," *Heat Transf.*, vol. n/a, no. n/a, Nov. 2022, doi: 10.1002/htj.22777.
- [27] J.-P. Hsu, J. Lou, Y.-Y. He, and E. Lee, "Diffusiophoresis of Concentrated Suspensions of Spherical Particles with Distinct Ionic Diffusion Velocities," *J. Phys. Chem. B*, vol. 111, no. 10, pp. 2533–2539, Mar. 2007, doi: 10.1021/jp0659305.
- [28] J. Lou, C.-Y. Shih, and E. Lee, "Diffusiophoresis of a spherical particle normal to an airwater interface," J. Colloid Interface Sci., vol. 331, no. 1, pp. 227–235, Mar. 2009, doi: 10.1016/j.jcis.2008.11.015.
- [29] J. L. Wilson, S. Shim, Y. E. Yu, A. Gupta, and H. A. Stone, "Diffusiophoresis in Multivalent Electrolytes," *Langmuir*, vol. 36, no. 25, pp. 7014–7020, Jun. 2020, doi: 10.1021/acs.langmuir.9b03333.
- [30] B. M. Alessio, S. Shim, E. Mintah, A. Gupta, and H. A. Stone, "Diffusiophoresis and diffusioosmosis in tandem: Two-dimensional particle motion in the presence of multiple electrolytes," *Phys. Rev. Fluids*, vol. 6, no. 5, p. 054201, May 2021, doi: 10.1103/PhysRevFluids.6.054201.
- [31] D. Florea, S. Musa, J. M. R. Huyghe, and H. M. Wyss, "Long-range repulsion of colloids driven by ion exchange and diffusiophoresis," *Proc. Natl. Acad. Sci.*, vol. 111, no. 18, pp. 6554–6559, May 2014, doi: 10.1073/pnas.1322857111.

- [32] H. A. Stone, A. D. Stroock, and A. Ajdari, "Engineering Flows in Small Devices: Microfluidics Toward a Lab-on-a-Chip," *Annu. Rev. Fluid Mech.*, vol. 36, no. 1, pp. 381– 411, Jan. 2004, doi: 10.1146/annurev.fluid.36.050802.122124.
- [33] T. J. Shimokusu, V. G. Maybruck, J. T. Ault, and S. Shin, "Colloid Separation by CO2-Induced Diffusiophoresis," *Langmuir*, vol. 36, no. 25, pp. 7032–7038, Dec. 2019, doi: 10.1021/acs.langmuir.9b03376.
- [34] S. Shim *et al.*, "CO<sub>2</sub> -Driven diffusiophoresis for maintaining a bacteria-free surface," *Soft Matter*, vol. 17, no. 9, pp. 2568–2576, 2021, doi: 10.1039/D0SM02023K.
- [35] A. Gupta, S. Shim, and H. A. Stone, "Diffusiophoresis: from dilute to concentrated electrolytes," *Soft Matter*, vol. 16, no. 30, pp. 6975–6984, 2020, doi: 10.1039/D0SM00899K.
- [36] K. Lee, J. Lee, D. Ha, M. Kim, and T. Kim, "Low-electric-potential-assisted diffusiophoresis for continuous separation of nanoparticles on a chip," *Lab. Chip*, vol. 20, no. 15, pp. 2735– 2747, 2020, doi: 10.1039/D0LC00196A.
- [37] H.-P. Hsu and E. Lee, "Electrophoresis of a single charged porous sphere in an infinite medium of electrolyte solution," *J. Colloid Interface Sci.*, vol. 390, no. 1, pp. 85–95, Jan. 2013, doi: 10.1016/j.jcis.2012.09.036.
- [38] E. Lee, "Diffusiophoresis of Porous Particles," in *Interface Science and Technology*, vol. 26, Elsevier, 2019, pp. 385–409. doi: 10.1016/B978-0-08-100865-2.00018-7.
- [39] W. Fang and E. Lee, "Diffusiophoretic motion of an isolated charged porous sphere," J. Colloid Interface Sci., vol. 459, pp. 273–283, Dec. 2015, doi: 10.1016/j.jcis.2015.08.002.
- [40] W. C. Li and H. J. Keh, "Diffusiophoretic mobility of charge-regulating porous particles: General," *ELECTROPHORESIS*, vol. 37, no. 15–16, pp. 2139–2146, Aug. 2016, doi: 10.1002/elps.201600091.
- [41] S.-C. Tsai and E. Lee, "Diffusiophoresis of a Highly Charged Porous Particle Induced by Diffusion Potential," *Langmuir*, vol. 35, no. 8, pp. 3143–3155, Feb. 2019, doi: 10.1021/acs.langmuir.8b04146.
- [42] H. C. Brinkman, "A calculation of the viscous force exerted by a flowing fluid on a dense swarm of particles," *Flow Turbul. Combust.*, vol. 1, no. 1, p. 27, Dec. 1949, doi: 10.1007/BF02120313.
- [43] H. Y. Huang and H. J. Keh, "Diffusiophoresis in Suspensions of Charged Porous Particles,"
  J. Phys. Chem. B, vol. 119, no. 5, pp. 2040–2050, Feb. 2015, doi: 10.1021/jp510448x.

- [44] J. L. Anderson, M. E. Lowell, and D. C. Prieve, "Motion of a particle generated by chemical gradients Part 1. Non-electrolytes," *J. Fluid Mech.*, vol. 117, pp. 107–121, Apr. 1982, doi: 10.1017/S0022112082001542.
- [45] D. C. Prieve, J. L. Anderson, J. P. Ebel, and M. E. Lowell, "Motion of a particle generated by chemical gradients. Part 2. Electrolytes," *J. Fluid Mech.*, vol. 148, pp. 247–269, Nov. 1984, doi: 10.1017/S0022112084002330.
- [46] B. V. Derjaguin, G. Sidorenkov, E. Zubashchenko, and E. Kiseleva, "Kinetic Phenomena in the boundary layers of liquids 1. the capillary osmosis," *Prog. Surf. Sci.*, vol. 43, no. 1, pp. 138–152, May 1993, doi: 10.1016/0079-6816(93)90023-O.
- [47] J. K. G. Dhont, S. Wiegand, S. Duhr, and D. Braun, "Thermodiffusion of Charged Colloids: Single-Particle Diffusion," *Langmuir*, vol. 23, no. 4, pp. 1674–1683, Feb. 2007, doi: 10.1021/la062184m.
- [48] B. V. Derjaguin, "Some results from 50 years' research on surface forces," in Surface Forces and Surfactant Systems, Darmstadt, 1987, pp. 17–30.
- [49] S. Marbach and L. Bocquet, "Osmosis, from molecular insights to large-scale applications," *Chem. Soc. Rev.*, vol. 48, no. 11, pp. 3102–3144, 2019, doi: 10.1039/C8CS00420J.
- [50] S. Vrhovec Hartman, B. Božič, and J. Derganc, "Migration of blood cells and phospholipid vesicles induced by concentration gradients in microcavities," *New Biotechnol.*, vol. 47, pp. 60–66, Dec. 2018, doi: 10.1016/j.nbt.2018.02.015.
- [51] H. Lee, J. Kim, J. Yang, S. W. Seo, and S. J. Kim, "Diffusiophoretic exclusion of colloidal particles for continuous water purification," *Lab. Chip*, vol. 18, no. 12, pp. 1713–1724, 2018, doi: 10.1039/C8LC00132D.
- [52] M. Seo, S. Park, D. Lee, H. Lee, and S. J. Kim, "Continuous and spontaneous nanoparticle separation by diffusiophoresis," *Lab. Chip*, vol. 20, no. 22, pp. 4118–4127, 2020, doi: 10.1039/D0LC00593B.
- [53] D. Lee and S. J. Kim, "Spontaneous diffusiophoretic separation in paper-based microfluidic device," *Micro Nano Syst. Lett.*, vol. 8, no. 1, p. 6, Dec. 2020, doi: 10.1186/s40486-020-00108-x.
- [54] S. Shin, P. B. Warren, and H. A. Stone, "Cleaning by Surfactant Gradients: Particulate Removal from Porous Materials and the Significance of Rinsing in Laundry Detergency,"

*Phys. Rev. Appl.*, vol. 9, no. 3, p. 034012, Mar. 2018, doi: 10.1103/PhysRevApplied.9.034012.

- [55] A. Kar, T.-Y. Chiang, I. Ortiz Rivera, A. Sen, and D. Velegol, "Enhanced Transport into and out of Dead-End Pores," ACS Nano, vol. 9, no. 1, pp. 746–753, Jan. 2015, doi: 10.1021/nn506216b.
- [56] D. C. Prieve, S. M. Malone, A. S. Khair, R. F. Stout, and M. Y. Kanj, "Diffusiophoresis of charged colloidal particles in the limit of very high salinity," *Proc. Natl. Acad. Sci.*, vol. 116, no. 37, pp. 18257–18262, Sep. 2019, doi: 10.1073/pnas.1701391115.
- [57] J. T. Ault, S. Shin, and H. A. Stone, "Diffusiophoresis in narrow channel flows," J. Fluid Mech., vol. 854, pp. 420–448, 2018, doi: 10.1017/jfm.2018.618.
- [58] D. Velegol, A. Garg, R. Guha, A. Kar, and M. Kumar, "Origins of concentration gradients for diffusiophoresis," *Soft Matter*, vol. 12, no. 21, pp. 4686–4703, 2016, doi: 10.1039/C6SM00052E.
- [59] G. M. Whitesides, "The origins and the future of microfluidics," *Nature*, vol. 442, no. 7101, pp. 368–373, Jul. 2006, doi: 10.1038/nature05058.
- [60] Q. Zhang and R. H. Austin, "Applications of Microfluidics in Stem Cell Biology," *BioNanoScience*, vol. 2, no. 4, pp. 277–286, Dec. 2012, doi: 10.1007/s12668-012-0051-8.
- [61] W.-H. Huang, F. Ai, Z.-L. Wang, and J.-K. Cheng, "Recent advances in single-cell analysis using capillary electrophoresis and microfluidic devices," *J. Chromatogr. B*, vol. 866, no. 1– 2, pp. 104–122, Apr. 2008, doi: 10.1016/j.jchromb.2008.01.030.
- [62] X. Ou, P. Chen, X. Huang, S. Li, and B. Liu, "Microfluidic chip electrophoresis for biochemical analysis," J. Sep. Sci., vol. 43, no. 1, pp. 258–270, Jan. 2020, doi: 10.1002/jssc.201900758.
- [63] R. Fernández-Mateo, V. Calero, H. Morgan, P. García-Sánchez, and A. Ramos, "Wall Repulsion of Charged Colloidal Particles during Electrophoresis in Microfluidic Channels," *Phys Rev Lett*, vol. 128, no. 7, p. 074501, Feb. 2022, doi: 10.1103/PhysRevLett.128.074501.
- [64] T. Tsuji, Y. Matsumoto, R. Kugimiya, K. Doi, and S. Kawano, "Separation of Nano- and Microparticle Flows Using Thermophoresis in Branched Microfluidic Channels," *Micromachines*, vol. 10, no. 5, p. 321, May 2019, doi: 10.3390/mi10050321.

- [65] T. Tsuji, K. Kozai, H. Ishino, and S. Kawano, "Direct observations of thermophoresis in microfluidic systems," *Micro Nano Lett.*, vol. 12, no. 8, pp. 520–525, Aug. 2017, doi: 10.1049/mnl.2017.0130.
- [66] N. Lee, S. Mohanakumar, and S. Wiegand, "Thermophoretic microfluidic cells for evaluating Soret coefficient of colloidal particles," *Int. J. Heat Mass Transf.*, vol. 194, p. 123002, Sep. 2022, doi: 10.1016/j.ijheatmasstransfer.2022.123002.
- [67] S. Battat, J. T. Ault, S. Shin, S. Khodaparast, and H. A. Stone, "Particle entrainment in deadend pores by diffusiophoresis," *Soft Matter*, vol. 15, no. 19, pp. 3879–3885, 2019, doi: 10.1039/C9SM00427K.
- [68] S. Shim, M. Baskaran, E. H. Thai, and H. A. Stone, "CO<sub>2</sub> -Driven diffusiophoresis and water cleaning: similarity solutions for predicting the exclusion zone in a channel flow," *Lab. Chip*, vol. 21, no. 17, pp. 3387–3400, 2021, doi: 10.1039/D1LC00211B.
- [69] S. Shim, "Diffusiophoresis, Diffusioosmosis, and Microfluidics: Surface-Flow-Driven Phenomena in the Presence of Flow," *Chem. Rev.*, vol. 122, no. 7, pp. 6986–7009, Apr. 2022, doi: 10.1021/acs.chemrev.1c00571.
- [70] S. Täuber, J. Schmitz, L. Blöbaum, N. Fante, H. Steinhoff, and A. Grünberger, "How to Perform a Microfluidic Cultivation Experiment—A Guideline to Success," *Biosensors*, vol. 11, no. 12, p. 485, Nov. 2021, doi: 10.3390/bios11120485.
- [71] D. C. Duffy, J. C. McDonald, O. J. A. Schueller, and G. M. Whitesides, "Rapid Prototyping of Microfluidic Systems in Poly(dimethylsiloxane)," *Anal. Chem.*, vol. 70, no. 23, pp. 4974– 4984, Dec. 1998, doi: 10.1021/ac980656z.
- [72] A. Gerami et al., "Microfluidics for Porous Systems: Fabrication, Microscopy and Applications," Transp. Porous Media, vol. 130, no. 1, pp. 277–304, Oct. 2019, doi: 10.1007/s11242-018-1202-3.
- [73] A. Jahanshahi, P. Salvo, and J. Vanfleteren, "PDMS Selective Bonding for the Fabrication of Biocompatible All Polymer NC Microvalves," *J. Microelectromechanical Syst.*, vol. 22, no. 6, pp. 1354–1360, Dec. 2013, doi: 10.1109/JMEMS.2013.2262595.
- [74] Shin-Won Kang, "Application of Soft Lithography for Nano Functional Devices," in *Lithography*, Michael Wang, Ed. Rijeka: IntechOpen, 2010, p. Ch. 20. doi: 10.5772/8186.

- [75] F. M. Wisser, B. Schumm, G. Mondin, J. Grothe, and S. Kaskel, "Precursor strategies for metallic nano- and micropatterns using soft lithography," *J. Mater. Chem. C*, vol. 3, no. 12, pp. 2717–2731, 2015, doi: 10.1039/C4TC02418D.
- [76] A. Anbari, H. Chien, S. S. Datta, W. Deng, D. A. Weitz, and J. Fan, "Microfluidic Model Porous Media: Fabrication and Applications," *Small*, vol. 14, no. 18, p. 1703575, May 2018, doi: 10.1002/smll.201703575.
- [77] P. Guillot *et al.*, "Viscosimeter on a Microfluidic Chip," *Langmuir*, vol. 22, no. 14, pp. 6438–6445, Jul. 2006, doi: 10.1021/la060131z.
- [78] M. J. Vold, "Zeta potential in colloid science. Principles and applications," J. Colloid Interface Sci., vol. 88, no. 2, p. 608, Aug. 1982, doi: 10.1016/0021-9797(82)90296-X.
- [79] B. Abécassis, C. Cottin-Bizonne, C. Ybert, A. Ajdari, and L. Bocquet, "Osmotic manipulation of particles for microfluidic applications," *New J. Phys.*, vol. 11, no. 7, p. 075022, Jul. 2009, doi: 10.1088/1367-2630/11/7/075022.
- [80] T.-Y. Chiang and D. Velegol, "Multi-ion diffusiophoresis," J. Colloid Interface Sci., vol. 424, pp. 120–123, Jun. 2014, doi: 10.1016/j.jcis.2014.03.003.
- [81] H. J. Keh, "Diffusiophoresis of charged particles and diffusioosmosis of electrolyte solutions," *Curr. Opin. Colloid Interface Sci.*, vol. 24, pp. 13–22, Aug. 2016, doi: 10.1016/j.cocis.2016.05.008.
- [82] X. Zhou et al., "Ionic Effects in Ionic Diffusiophoresis in Chemically Driven Active Colloids," Phys Rev Lett, vol. 127, no. 16, p. 168001, Oct. 2021, doi: 10.1103/PhysRevLett.127.168001.
- [83] B. Abécassis, C. Cottin-Bizonne, C. Ybert, A. Ajdari, and L. Bocquet, "Boosting migration of large particles by solute contrasts," *Nat. Mater.*, vol. 7, no. 10, pp. 785–789, Oct. 2008, doi: 10.1038/nmat2254.
- [84] S. Shin, "Diffusiophoretic separation of colloids in microfluidic flows," *Phys. Fluids*, vol. 32, no. 10, p. 101302, Oct. 2020, doi: 10.1063/5.0023415.
- [85] S. Shin *et al.*, "Size-dependent control of colloid transport via solute gradients in dead-end channels," *Proc. Natl. Acad. Sci.*, vol. 113, no. 2, pp. 257–261, Jan. 2016, doi: 10.1073/pnas.1511484112.

- [86] V. S. Doan, P. Saingam, T. Yan, and S. Shin, "A Trace Amount of Surfactants Enables Diffusiophoretic Swimming of Bacteria," ACS Nano, vol. 14, no. 10, pp. 14219–14227, Oct. 2020, doi: 10.1021/acsnano.0c07502.
- [87] A. Banerjee, I. Williams, R. N. Azevedo, M. E. Helgeson, and T. M. Squires, "Soluto-inertial phenomena: Designing long-range, long-lasting, surface-specific interactions in suspensions," *Proc. Natl. Acad. Sci.*, vol. 113, no. 31, pp. 8612–8617, Aug. 2016, doi: 10.1073/pnas.1604743113.
- [88] S. Musa, D. Florea, H. M. Wyss, and J. M. Huyghe, "Convection associated with exclusion zone formation in colloidal suspensions," *Soft Matter*, vol. 12, no. 4, pp. 1127–1132, 2016, doi: 10.1039/C5SM01502B.
- [89] Y. Cheng and C. I. Moraru, "Long-range interactions keep bacterial cells from liquid-solid interfaces: Evidence of a bacteria exclusion zone near Nafion surfaces and possible implications for bacterial attachment," *Colloids Surf. B Biointerfaces*, vol. 162, pp. 16–24, Feb. 2018, doi: 10.1016/j.colsurfb.2017.11.016.
- [90] B. J. Kirby and E. F. Hasselbrink, "Zeta potential of microfluidic substrates: 1. Theory, experimental techniques, and effects on separations," *ELECTROPHORESIS*, vol. 25, no. 2, pp. 187–202, Jan. 2004, doi: 10.1002/elps.200305754.
- [91] R.-Y. Chen, "Flow in the Entrance Region at Low Reynolds Numbers," *J. Fluids Eng.*, vol. 95, no. 1, pp. 153–158, Mar. 1973, doi: 10.1115/1.3446948.
- [92] F. Durst, S. Ray, B. Ünsal, and O. A. Bayoumi, "The Development Lengths of Laminar Pipe and Channel Flows," J. Fluids Eng., vol. 127, no. 6, pp. 1154–1160, Nov. 2005, doi: 10.1115/1.2063088.
- [93] S.-Y. Lee, J. Jang, and S. T. Wereley, "Effects of planar inlet plenums on the hydrodynamically developing flows in rectangular microchannels of complementary aspect ratios," *Microfluid. Nanofluidics*, vol. 5, no. 1, pp. 1–12, Jul. 2008, doi: 10.1007/s10404-007-0179-y.
- [94] T. Ahmad and I. Hassan, "Experimental Analysis of Microchannel Entrance Length Characteristics Using Microparticle Image Velocimetry," J. Fluids Eng., vol. 132, no. 4, p. 041102, Apr. 2010, doi: 10.1115/1.4001292.
- [95] O. J. Lobo and D. Chatterjee, "Development of flow in a square mini-channel: Effect of flow oscillation," *Phys. Fluids*, vol. 30, no. 4, p. 042003, Apr. 2018, doi: 10.1063/1.5018160.

- [96] S. Shim and H. A. Stone, "CO 2 -leakage-driven diffusiophoresis causes spontaneous accumulation of charged materials in channel flow," *Proc. Natl. Acad. Sci.*, vol. 117, no. 42, pp. 25985–25990, Oct. 2020, doi: 10.1073/pnas.2010011117.
- [97] S. Shim, J. Wan, S. Hilgenfeldt, P. D. Panchal, and H. A. Stone, "Dissolution without disappearing: multicomponent gas exchange for CO2 bubbles in a microfluidic channel," *Lab. Chip*, vol. 14, no. 14, p. 2428, 2014, doi: 10.1039/c4lc00354c.
- [98] Y. K. Wei and H. J. Keh, "Diffusiophoresis in a Suspension of Spherical Particles with Arbitrary Double-Layer Thickness," J. Colloid Interface Sci., vol. 248, no. 1, pp. 76–87, Apr. 2002, doi: 10.1006/jcis.2001.8175.
- [99] J. Lou and E. Lee, "Diffusiophoresis of a Spherical Particle Normal to a Plane," J. Phys. Chem. C, vol. 112, no. 7, pp. 2584–2592, Feb. 2008, doi: 10.1021/jp076586u.
- [100] P. Y. Huang and H. J. Keh, "Diffusiophoresis of a Spherical Soft Particle in Electrolyte Gradients," J. Phys. Chem. B, vol. 116, no. 25, pp. 7575–7589, Jun. 2012, doi: 10.1021/jp302836g.
- [101] J. Lou, C.-Y. Shih, and E. Lee, "Diffusiophoresis of Concentrated Suspensions of Spherical Particles with Charge-regulated Surface: Polarization Effect with Nonlinear Poisson–Boltzmann Equation," *Langmuir*, vol. 26, no. 1, pp. 47–55, Jan. 2010, doi: 10.1021/la902113s.
- [102] J. TH. G. OVERBEEK, "POLYELECTROLYTES, PAST, PRESENT AND FUTURE," in *Macromolecular Chemistry–11*, H. Eisenberg, Ed. Pergamon, 1977, pp. 91–101. doi: 10.1016/B978-0-08-020975-3.50004-9.
- [103] J. T. Mannion and H. G. Craighead, "Nanofluidic structures for single biomolecule fluorescent detection," *Biopolymers*, vol. 85, no. 2, pp. 131–143, Feb. 2007, doi: 10.1002/bip.20629.
- [104] B. M. Al-Shehri, A. E. R. S. Khder, S. S. Ashour, and M. S. Hamdy, "A review: the utilization of mesoporous materials in wastewater treatment," *Mater. Res. Express*, vol. 6, no. 12, p. 122002, Nov. 2019, doi: 10.1088/2053-1591/ab52af.
- [105] E. Lee, "Electrophoresis of Porous Particles," in *Interface Science and Technology*, vol. 26, Elsevier, 2019, pp. 145–180. doi: 10.1016/B978-0-08-100865-2.00007-2.
- [106] E. Lee, "Electrophoresis of a Single Rigid Particle," in *Interface Science and Technology*, vol. 26, Elsevier, 2019, pp. 3–45. doi: 10.1016/B978-0-08-100865-2.00001-1.

- [107] E. Lee, "Diffusiophoresis of Rigid Particles," in *Interface Science and Technology*, vol. 26, Elsevier, 2019, pp. 323–358. doi: 10.1016/B978-0-08-100865-2.00016-3.
- [108] R. Byron Bird, Warren E. Stewart, and Edwin N. Lightfoot, *Transport Phenomena*, Second Edition. John Wiley & sons, 2007.
- [109] I. Klyuzhin, A. Symonds, J. Magula, and G. H. Pollack, "New Method of Water Purification Based on the Particle-Exclusion Phenomenon," *Environ. Sci. Technol.*, vol. 42, no. 16, pp. 6160–6166, Aug. 2008, doi: 10.1021/es703159q.
- [110] H. Lee, J. Kim, J. Yang, S. W. Seo, and S. J. Kim, "Diffusiophoretic exclusion of colloidal particles for continuous water purification," *Lab. Chip*, vol. 18, no. 12, pp. 1713–1724, 2018, doi: 10.1039/C8LC00132D.
- [111] S. Shin, O. Shardt, P. B. Warren, and H. A. Stone, "Membraneless water filtration using CO2," *Nat. Commun.*, vol. 8, no. 1, p. 15181, Aug. 2017, doi: 10.1038/ncomms15181.
- [112] J. S. Paustian, R. N. Azevedo, S.-T. B. Lundin, M. J. Gilkey, and T. M. Squires, "Microfluidic Microdialysis: Spatiotemporal Control over Solution Microenvironments Using Integrated Hydrogel Membrane Microwindows," *Phys. Rev. X*, vol. 3, no. 4, p. 041010, Nov. 2013, doi: 10.1103/PhysRevX.3.041010.
- [113] R. Pedregal-Cortés, G. Toriz, E. Delgado, and G. H. Pollack, "Interfacial water and its potential role in the function of sericin against biofouling," *Biofouling*, vol. 35, no. 7, pp. 732–741, Aug. 2019, doi: 10.1080/08927014.2019.1653863.
- [114] R. Niu *et al.*, "Controlled assembly of single colloidal crystals using electro-osmotic micropumps," *Phys. Chem. Chem. Phys.*, vol. 19, no. 4, pp. 3104–3114, 2017, doi: 10.1039/C6CP07231C.
- [115] B. Liebchen, R. Niu, T. Palberg, and H. Löwen, "Unraveling modular microswimmers: From self-assembly to ion-exchange-driven motors," *Phys. Rev. E*, vol. 98, no. 5, p. 052610, Nov. 2018, doi: 10.1103/PhysRevE.98.052610.
- [116] J. Zheng and G. H. Pollack, "Long-range forces extending from polymer-gel surfaces," *Phys. Rev. E*, vol. 68, no. 3, p. 031408, Sep. 2003, doi: 10.1103/PhysRevE.68.031408.
- [117] B. V. Derjaguin and M. V. Golovanov, "On long-range forces of repulsion between biological cells," *Phys. Chem. Colloids Interfaces Biotechnol. Drug Res.*, vol. 10, pp. 77–84, Jan. 1984, doi: 10.1016/0166-6622(84)80009-8.

- [118] B. Chai, A. G. Mahtani, and G. H. Pollack, "UNEXPECTED PRESENCE OF SOLUTE-FREE ZONES AT METAL-WATER INTERFACES," *Contemp. Mater.*, vol. 1, no. 3, pp. 1–12, Oct. 2012, doi: 10.7251/COM1201001C.
- [119] M. J. Esplandiu, D. Reguera, and J. Fraxedas, "Electrophoretic origin of long-range repulsion of colloids near water/Nafion interfaces," *Soft Matter*, vol. 16, no. 15, pp. 3717– 3726, 2020, doi: 10.1039/D0SM00170H.
- [120] A. V. Delgado, F. González-Caballero, R. J. Hunter, L. K. Koopal, and J. Lyklema, "Measurement and Interpretation of Electrokinetic Phenomena (IUPAC Technical Report)," *Pure Appl. Chem.*, vol. 77, no. 10, pp. 1753–1805, 2005, doi: 10.1351/pac200577101753.
- [121] B. Chai, H. Yoo, and G. H. Pollack, "Effect of Radiant Energy on Near-Surface Water," J. *Phys. Chem. B*, vol. 113, no. 42, pp. 13953–13958, Oct. 2009, doi: 10.1021/jp908163w.
- [122] J. Zheng, W.-C. Chin, E. Khijniak, E. Khijniak, and G. H. Pollack, "Surfaces and interfacial water: Evidence that hydrophilic surfaces have long-range impact," *Adv. Colloid Interface Sci.*, vol. 127, no. 1, pp. 19–27, Nov. 2006, doi: 10.1016/j.cis.2006.07.002.
- [123] J. Zheng, A. Wexler, and G. H. Pollack, "Effect of buffers on aqueous solute-exclusion zones around ion-exchange resins," *J. Colloid Interface Sci.*, vol. 332, no. 2, pp. 511–514, Apr. 2009, doi: 10.1016/j.jcis.2009.01.010.
- [124] C.-S. Chen, W.-J. Chung, I. C. Hsu, C.-M. Wu, and W.-C. Chin, "Force field measurements within the exclusion zone of water," *J. Biol. Phys.*, vol. 38, no. 1, pp. 113–120, Jan. 2012, doi: 10.1007/s10867-011-9237-5.
- [125] I. Huszár, Z. Mártonfalvi, A. Laki, K. Iván, and M. Kellermayer, "Exclusion-Zone Dynamics Explored with Microfluidics and Optical Tweezers," *Entropy*, vol. 16, no. 8, pp. 4322–4337, Aug. 2014, doi: 10.3390/e16084322.
- [126] V. A. Papadimitriou, L. I. Segerink, and J. C. T. Eijkel, "Continuous focusing, fractionation and extraction of anionic analytes in a microfluidic chip," *Lab. Chip*, vol. 19, no. 19, pp. 3238–3248, 2019, doi: 10.1039/C9LC00434C.
- [127] J. T. Ault, P. B. Warren, S. Shin, and H. A. Stone, "Diffusiophoresis in one-dimensional solute gradients," *Soft Matter*, vol. 13, no. 47, pp. 9015–9023, 2017, doi: 10.1039/C7SM01588G.

- [128] Z. Almutairi, C. L. Ren, and L. Simon, "Evaluation of polydimethylsiloxane (PDMS) surface modification approaches for microfluidic applications," *Colloids Surf. Physicochem. Eng. Asp.*, vol. 415, pp. 406–412, Dec. 2012, doi: 10.1016/j.colsurfa.2012.10.008.
- [129] Jacob H. Masliyah and Subir Bhattacharjee, *Electrokinetic and Colloid Transport Phenomena*. John Wiley & Sons, 2006.
- [130] C.-S. Chen, E. Farr, J. Anaya, E. Chen, and W.-C. Chin, "The Solute-Exclusion Zone: A Promising Application for Mirofluidics," *Entropy*, vol. 17, no. 3, pp. 1466–1476, Mar. 2015, doi: 10.3390/e17031466.
- [131] D. C. Prieve and R. Roman, "Diffusiophoresis of a rigid sphere through a viscous electrolyte solution," J. Chem. Soc. Faraday Trans. 2, vol. 83, no. 8, p. 1287, 1987, doi: 10.1039/f29878301287.
- [132] H. J. Keh and Y. K. Wei, "Diffusiophoretic Mobility of Spherical Particles at Low Potential and Arbitrary Double-Layer Thickness," *Langmuir*, vol. 16, no. 12, pp. 5289–5294, Jun. 2000, doi: 10.1021/la991373k.
- [133] A. S. Khair, "Diffusiophoresis of colloidal particles in neutral solute gradients at finite Péclet number," J. Fluid Mech., vol. 731, pp. 64–94, Sep. 2013, doi: 10.1017/jfm.2013.364.
- [134] A. Gupta, B. Rallabandi, and H. A. Stone, "Diffusiophoretic and diffusioosmotic velocities for mixtures of valence-asymmetric electrolytes," *Phys. Rev. Fluids*, vol. 4, no. 4, p. 043702, Apr. 2019, doi: 10.1103/PhysRevFluids.4.043702.
- [135] O. Annunziata, D. Buzatu, and J. G. Albright, "Protein Diffusiophoresis and Salt Osmotic Diffusion in Aqueous Solutions," J. Phys. Chem. B, vol. 116, no. 42, pp. 12694–12705, Oct. 2012, doi: 10.1021/jp307625d.
- [136] M. S. McAfee and O. Annunziata, "Effects of Salting-In Interactions on Macromolecule Diffusiophoresis and Salt Osmotic Diffusion," *Langmuir*, vol. 31, no. 4, pp. 1353–1361, Feb. 2015, doi: 10.1021/la5046223.
- [137] I. Miranda *et al.*, "Properties and Applications of PDMS for Biomedical Engineering: A Review," *J. Funct. Biomater.*, vol. 13, no. 1, p. 2, Dec. 2021, doi: 10.3390/jfb13010002.
- [138] B. J. Kirby and E. F. Hasselbrink, "Zeta potential of microfluidic substrates: 2. Data for polymers," *ELECTROPHORESIS*, vol. 25, no. 2, pp. 203–213, Jan. 2004, doi: 10.1002/elps.200305755.

- [139] S. Shim, J. K. Nunes, G. Chen, and H. A. Stone, "Diffusiophoresis in the presence of a pH gradient." arXiv, Sep. 28, 2022. Accessed: Dec. 02, 2022. [Online]. Available: http://arxiv.org/abs/2209.14442
- [140] G. S. Manning, "Limiting Laws and Counterion Condensation in Polyelectrolyte Solutions II. Self-Diffusion of the Small Ions," *J. Chem. Phys.*, vol. 51, no. 3, pp. 934–938, Aug. 1969, doi: 10.1063/1.1672158.
- [141] H.-P. Hsu and E. Lee, "Counterion condensation of a polyelectrolyte," *Electrochem. Commun.*, vol. 15, no. 1, pp. 59–62, Feb. 2012, doi: 10.1016/j.elecom.2011.11.024.
- [142] L.-H. Yeh and J.-P. Hsu, "Effects of double-layer polarization and counterion condensation on the electrophoresis of polyelectrolytes," *Soft Matter*, vol. 7, no. 2, pp. 396–411, 2011, doi: 10.1039/C0SM00600A.
- [143] R. J. HUNTER, "Chapter 2 Charge and Potential Distribution at Interfaces," in Zeta Potential in Colloid Science, R. J. HUNTER, Ed. Academic Press, 1981, pp. 11–58. doi: 10.1016/B978-0-12-361961-7.50006-7.